Contents lists available at ScienceDirect

Renewable and Sustainable Energy Reviews

journal homepage: www.elsevier.com/locate/rser

Review article

Defining characteristics of peer-to-peer energy trading, transactive energy, and community self-consumption: A review of literature and expert perspectives

Anna Gorbatcheva^{*,1}, Nicole Watson¹, Alexandra Schneiders, David Shipworth, Michael J. Fell Energy Institute, University College London, 14 Upper Woburn Place, London, WC1H ONN, United Kingdom

ARTICLE INFO

ABSTRACT

Dataset link: https://osf.io/mn7zk/

Keywords: Peer-to-peer Community self-consumption Collective self-consumption Transactive energy Energy trading Local energy market

To facilitate a successful integration of distributed energy resources into the electricity generation mix, new forms of energy markets must be considered. Concepts such as Peer-to-peer energy trading (P2P), transactive energy (TE) and community/collective self-consumption (CSC) are frequently mentioned as solutions to this challenge. Despite increasing interest from industry, policy, and academia, the field lacks a shared understanding of this class of models. This need is addressed by presenting sets of shared and distinct characteristics which define P2P, TE and CSC. Our analysis is based on a series of expert group interviews with regulators, industry, and academics across 13 countries, and a systematic and targeted literature review of 133 papers. Findings show that P2P/TE/CSC models can be described as sub-markets that operate within or alongside traditional energy markets and enable trading or sharing of energy using an automated approach. They focus on promoting and supporting local energy generation and consumption using price negotiation mechanisms that reflect the aims of the market. The paper also presents sets of characteristics which differentiate P2P, TE, and CSC from one another and sets out guiding definitions to be used as a reference point. The main differences between these models stem from the goal they are trying to achieve and the contexts they are deployed in. Findings from this analysis can support development of a shared understanding of this class of models across multiple disciplinary perspectives and applications.

Contents

1.	Introdu	uction		2
2.	Metho	dological a	approach	3
	2.1.	Expert in	nterviews	3
	2.2.	Literatur	e review	4
3.	Definii	ng the clas	s of models: Expert interview results	4
4.	Differe	ntiation o	f models: Literature review findings	7
	4.1.	Qualitati	ve content analysis	7
	4.2.	Analysis	of terms	7
	4.3.	Analysis	of characteristics	8
		4.3.1.	Market or system aims and incentives	9
		4.3.2.	Participant type and scale	10
		4.3.3.	Participant interactions	10
		4.3.4.	Market or system design	10
		4.3.5.	Market or system scale	11
		4.3.6.	Market transactions and settlement	12
		4.3.7.	Governance and policy	12
5.	Discus	sion		13
	5.1.	Common	themes	13

* Corresponding author.

https://doi.org/10.1016/j.rser.2024.114672

Received 10 March 2023; Received in revised form 15 April 2024; Accepted 14 June 2024 Available online 1 July 2024

1364-0321/© 2024 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).







E-mail address: anna.gorbatcheva.17@ucl.ac.uk (A. Gorbatcheva).

 $^{^{1}\,}$ Nicole Watson and Anna Gorbatcheva have contributed equally to this work and are joint first author.

		5.1.1.	Digitalisation and automation	. 13				
		5.1.2.	Values	. 13				
		5.1.3.	Scale and governance	. 13				
		5.1.4.	Market design	13				
	5.2.	Compari	son between models	. 14				
		5.2.1.	P2P energy trading	. 14				
		5.2.2.	Transactive energy	. 14				
		5.2.3.	Collective/community self-consumption	. 14				
6.	Conclu	sion		16				
	Declara	ation of c	ompeting interest	. 17				
	Data a	vailability		17				
	Acknow	wledgeme	nts	17				
	Append	dix A. Ov	erview of characteristics and references	. 17				
Appendix B. Differentiation tool for P2P/TE/CSC models								
	Referen	nces		18				

Nomenclature

API	Application programming interface
CSC	Community or collective self-consumption
DER	Distributed energy resource
DLT	Distributed ledger technology
GO-P2P	Global Observatory on Peer-to-Peer, Com- munity Self-Consumption and Transactive Energy Models
LEM	Local energy market
LV	Low-voltage
P2P	Peer-to-peer
PV	Photovoltaic
RED II	Renewable Energy Directive II
RES	Renewable energy source
SI	Similarity index
TE	Transactive energy
VPP	Virtual power plant

1. Introduction

The increasing deployment of Distributed Energy Resources (DERs) to meet carbon reduction targets poses enormous challenges to the traditional energy supply system, which has been previously characterised by few centralised generation units with high reliability. Focus has shifted towards decentralised energy management, given the increasing availability of decentralised renewable generation units, storage facilities, and electric vehicles. As energy end users become more active in their generation and consumption behaviour, their traditionally passive role in the energy market must be reconsidered [1]. Currently, domestic consumers with DER assets, known as 'prosumers', do not have a platform for active participation in energy services, meaning that they remain an untapped source of potential for future energy systems [2].

In response, decentralised or local energy markets (LEMs) have been proposed in academic literature and trialled in real-world pilot projects [3,4]. Regulators and policymakers globally are working with tools such as regulatory sandboxes [5] to understand how these markets can be incorporated and harmonised with existing energy market structures [6,7]. The challenge is to democratise and decentralise energy markets by allowing new market participants to contribute whilst at the same time ensuring system stability and reliability. Commonly referred to models of LEMs include, but are not limited to, Peer-to-Peer Energy Trading (P2P, or 'peer-to-peer'), Transactive Energy (TE) and Community or Collective Self-Consumption (CSC). These models share similar underlying concepts but have differences in their key objectives. Regulation of these models must balance creating a supportive environment for innovation with ensuring consumer protection. There are several unresolved regulatory barriers in this field, such as the 'single supplier model', challenges related to data sharing, and the lack of defined rights and responsibilities for prosumers (see [7] for comprehensive discussion).

Despite the industry, policy, and academic attention P2P, TE and CSC models have attracted, the field lacks a shared definition of this class of models. The research presented in this paper was conducted as part of the Global Observatory on Peer-to-Peer, Community Self-Consumption and Transactive Energy Models (GO-P2P), a task of the User-Centred Energy Systems Technology Collaboration Programme (Users TCP), which runs under the auspices of the IEA (International Energy Agency). GO-P2P is a forum for international collaboration for academic experts, industry innovators, policymakers and other stakeholders across multiple disciplinary backgrounds, which aims to understand the policy, regulatory, social and technological conditions necessary to support the wider deployment of P2P, TE and CSC models. The key aim of GO-P2P is to collect data from pilots of P2P, TE and CSC and run an international comparative analysis to find out what are the enabling and inhibiting factors for the rollout of these models. To understand which pilots to include in this work and to categorise them, it is necessary to understand the defining characteristics of P2P, TE, and CSC [8].

Knowledge gained from GO-P2P shows that terms are often used interchangeably or inconsistently across different pilot projects, geographical locations, and policy contexts. This condition partially stems from the nascent and highly applied nature of this field: Firstly, this is an evolving field and the specific time period in which a term originated can influence how it is understood as technologies and knowledge develop. Similarly, where a term has originated in relation to a particular location it can guide how a term is used and understood in the particular regional or national context, particularly if academic researchers or innovators attempt to align their work with current policy priorities. Finally, this is a cross-disciplinary field: the disciplinary perspective from which a term originates can impact the main focus of its definition. The combination of the three factors – time period, geographical location, and discipline – can result in a multiplicity of definitions that might or might not refer to the same concept.

A shared understanding of these concepts is a basic prerequisite to allow knowledge exchange between industry and academia, avoid misconception and encourage synergy effects. There have been attempts at providing definitions for P2P, TE and CSC, usually led by larger organisations in the field [9,10]. Appropriate concept definition is essential for policy-making, given that countries are already starting to draft legislation enabling P2P, TE and CSC models to be developed (e.g., [9]). This will enable collaboration between sectors, as well as exchange of best practices at national and international level. Nonetheless, most definitions focus on a single aspect of the concepts, rather than a holistic examination of all aspects of these models. There have already been several comprehensive systematic reviews P2P, TE, and CSC systems [7,11-14]. However, these reviews aimed to identify evidence gaps from a particular perspective, rather than to bring together a holistic understanding of these models.

This paper meets this need by addressing the following research question: What are the key defining characteristics of energy systems labelled as Peer-to-Peer Energy Trading, Community/Collective Self-Consumption or Transactive Energy as outlined in the literature and by stakeholders? To answer this question, the following aims are addressed:

- 1. To understand which characteristics distinguish P2P, TE, CSC models from traditional energy systems.
- 2. To understand which broader terms are being used in the literature to refer to P2P, TE, and CSC-type models and how similar they are to one another with regards to the number of characteristics they share.
- To understand which characteristics are associated with P2P, TE, and CSC respectively, and where the similarities and differences between them lie.
- 4. To reflect critically on how the terms P2P, TE, and CSC are being used and understand implications for the sector.

Given the breadth and evolving nature of this field, providing a comprehensive review of all definitions is challenging. Instead, this paper brings together multiple data sources to reach an informed and shared understanding of how these terms are being used across different disciplines and contexts. Five expert group interviews with academics, industry, and policy representatives in 13 countries were conducted to identify key defining characteristics that distinguish P2P, TE and CSC models from traditional energy systems. This was followed by systematic and targeted reviews of academic and grey literature to determine how these terms are used in the field, and identified characteristics to distinguish the three models from one another.

The paper is structured as follows: Section 2 describes the overarching methodology of the paper including the data collection process for expert interviews and literature review. Section 3 presents results from the expert interviews, which aimed to understand which features a pilot would have to exhibit to belong to this class of models. Section 4 presents results from the systematic and targeted reviews, which aimed to identify characteristics that distinguish P2P, TE, and CSC from one another. Section 5 discusses findings and presents guiding definitions. Section 6 concludes by offering a future outlook and recommendations.

2. Methodological approach

In this paper, semi-structured expert group interviews with a systematic and targeted review of academic and grey literature are combined, in order to capture diverse perspectives and allow the strengths of each method to overcome the limitations of the other. Specific methods are outlined in the subsections below.

This approach borrows from the field of terminology science, which is concerned with how terms develop and evolve in practice. It specifically borrows from 'textual' (also called 'corpus terminology') and 'socioterminological' approaches [15]. The ISO/TR 22134:2007 'Practical Guidelines For Socioterminology' defines socioterminology as an "... approach of terminology work based on the sociological, cultural and sociolinguistic characteristics of a linguistic community, aiming at the study and the development of its technolects in accordance with those characteristics". In other words, socioterminology looks to understand the technical language of a subject or field by examining various aspects of its use by the community's experts. Socioterminology begins with an analysis of discourse within the field of study (either written or spoken) to construct "textual corpuses corresponding to the various scientific, technical and professional communication situations". These corpuses are then analysed to understand discursive practices relating to how terms are used within the field [16]. This is reflected in the expert group interview work conducted in this paper. Textual, or corpus,

Table 1

Number of interview participants and participating countries from each thematic group.

Sub-taks expert group	Number of participants	Number of countries
Sub-task 1 group: Power systems integration	20	6
Sub-task 2 group: ICT and data	8	5
Sub-task 3 group: Transactions and markets	21	8
Sub-task 4 group: Social and economic value	17	9
Sub-task 5 group: Policy and regulation	16	9

terminology "consists of examining the corpus of texts produced freely by experts in a field in order to extract from it significant factors that enable the concepts to be reconstructed and to extract the data essential for drafting definitions, and terms or "candidate terms" that designate the concepts" [16]. This approach is reflected in the literature analysis conducted in this paper.

While some of the terms (P2P, CSC and TE) had definitions created within specific disciplinary and geographic domains (and these were included in the corpus terminology review), there were widely differing uses of the terms between disciplines and countries. Given the nascent nature of the terms (P2P, CSC and TE) and the differing use in practice across disciplinary and geographic domains, a primarily socioterminological approach of mapping how the terms are used in practice was adopted looking for shared characteristics between terms as a constructive step on the road to shared understanding across disciplines. The steps followed in this paper conform to the European Commission's Working Party on Terminology and Documentation Recommendations for Terminology Work (2nd edition) 2003. These guidelines recommend a combination of analysis of specialist documents and the development of terms in conjunction with expert opinions.

2.1. Expert interviews

Expert interviews aim to elicit specific knowledge from individuals with specialised knowledge on the topic [17]. Expert interviews were used for the purpose of identifying a set of characteristics that distinguish P2P/TE/CSC models from traditional energy systems.

Experts from a wide range of disciplines and sectors in 13 countries across four continents were interviewed: Europe, North America, South America and Australia. Expert interviewees were recruited from GO-P2P. Five expert group interviews were conducted in the summer of 2020, each focusing on a different aspect of the class of models encompassing P2P/TE/CSC. This enabled a cross-disciplinary perspective on understanding these models. Interviewees consisted of academics, regulators, and industry stakeholders. Table 1 outlines the sub-tasks corresponding to GO-P2P's working groups.

Prior to the interview, all participants were asked to submit their top five key characteristics they believed distinguish these models from traditional energy systems from the perspective of their group:

"Please list a minimum of five key common characteristics of Peerto-Peer Energy Trading (P2P), Community/Collective Self-Consumption (CSC) and Transactive Energy (TE) models from the perspective of subtask 3 (transactions and markets). From this perspective, what are the main characteristics that are shared by P2P, TE and CSC models?

By characteristic, we mean a feature or quality belonging typically to a person, place, or thing and serving to identify them. Please focus on attributes that describe P2P/CSC/TE models using single words or phrases rather than full sentences".

During the interview, participants were asked to openly comment on the list of submitted characteristics and add additional characteristics, before voting on the top defining characteristics of this class of



Fig. 1. Overview of expert interview analysis method.

model from their group's perspective. A semi-structured discussion was held around the outcome of the vote until consensus was reached on the top defining characteristics. After the interview, participants were asked to individually rank order the top characteristics. A weighted average score for each characteristic was calculated, with respondents' first choice having a weight of eight, second choice having a weight of seven, and so on. Characteristics were coded thematically by a single independent coder. Fig. 1 shows an overview of the steps taken before, during, and after the interview. Please see supplementary material for the full interview guide.

Group (as opposed to individual) interviews allowed a wider range of views to be collected, as well as identifying points of consensus and disagreement within the field.

As there was a risk that country or context specific terms could dominate the expert interviews, the interviews focused on the defining characteristics of the *class of models*, rather than specific terms. Another challenge posed by interviewing stakeholders with an interest in this field is the possibility of introducing bias, for example stakeholders suggesting defining characteristics aligned with their own projects.

2.2. Literature review

The literature review counters the challenge of potential bias in the interviews by offering insight into how the terms P2P, TE and CSC are each used in a sample of academic and industry literature. It also allows us to understand how these terms are used in the field, outside of an interview situation where participants are aware that they are being observed.

Data took the form of definitions and characteristics of P2P, TE and CSC models found in a sample of academic and non-academic (i.e. industry and policy) literature. Literature was identified through a combination of systematic and targeted approaches. First, a pre-registered, systematic search was run using combinations of the terms *peer-topeer/peer-to-peer energy trading, community/collective self-consumption and transactive energy* + '*defin**'. The search was conducted in August 2020 and the full search protocol can be found in the supplementary material or at https://osf.io/bnfmp/.

During screening, it was noted that the academic literature identified by this protocol was dominated by modelling papers focused on understanding the effect of different market parameters. To achieve more equal representation of literature across disciplines and ensure that key papers had not been missed out, an additional call for literature specifically discussing definitions or characteristics of P2P/TE/CSC models was issued to the GO-P2P community. This resulted in additional papers being added (both academic and non-academic). Duplicates were removed and papers were screened using criteria applied in the systematic review. Given that much of the work in this field is more amenable to study by simulation rather than more resource-intensive field trials, a frequentist sampling of the literature was not undertaken. This allowed us to focus on characteristics that are present or absent for each model rather than replicating the state of evolution of the field. Combining the literature review with the expert interviews also helps overcome this limitation of the literature review, as the interviews represent a wide range of disciplinary perspectives.

Out of all papers, 78 academic and 55 non-academic papers contained either definitions or characteristics of P2P, TE, or CSC and were included in the final analysis. Fig. 2 shows the number of papers identified at each stage of the screening process.

The literature was restricted to articles only published in English, meaning that some contexts outside of English-speaking countries may have been missed. However, the wide geographical representation provided by interviewees mitigates this limitation.

Finally, data from the expert interviews and the literature review was triangulated, drawing from both sources to critically reflect on how the terms P2P, TE, and CSC are being used among experts and in literature and understand its implications for the energy sector.

3. Defining the class of models: Expert interview results

The following sections present the results of the expert interviews aimed at understanding which characteristics distinguish P2P, TE and CSC models from traditional energy systems. Fig. 3 shows an overview of the characteristics identified by each thematic group as the top, defining characteristics of this class of models.



Fig. 2. Overview of literature review data collection.

In the group focused on Power systems integration, eight defining characteristics of the class of models encompassing P2P, TE and CSC were identified. According to this group, for a system to be considered as belonging to this class of models, it must: support integration of local generation; be decentralised; use local distribution networks (i.e. low voltage (LV) networks); include a smart energy interface and control mechanism; participants must have smart meters; the system must support the integration of renewable energy; promote energy selfconsumption; and operate autonomously. During the discussion, the importance of locality as a defining characteristic of these models was highlighted, although participants noted that the community-orientated nature of these models had been missed in the characteristics put forward. Interview participants were of the view that the level of complexity involved in P2P/TE/CSC energy trading is too great to allow participants to make manual transactions and that consumers are unable to make informed decisions about aspects such as active/reactive power. For these reasons, it was felt that an autonomous system would be required.

The ICT and data group identified nine defining characteristics of P2P/TE/CSC models. These were: advanced metering infrastructure; interoperable data that can be shared with and interpreted by all actors in the system; components of the system have application programming interfaces allowing for the exchange of data and control commands between system components; distributed infrastructure for data management; consumer interfaces that allow participants to express trading preferences; devices have a traceable identity; independent agents at device level; compliance with local data protection laws; and ICT having the capacity to deal with current and future frequency, diversity and volume of data. Notably, many of the characteristics described by this group focus more on practical elements required for the system to operate, rather than features that necessarily distinguish the system from traditional energy systems. The topic of interoperability played a key role both in terms of data structures and devices. Similarly to the Power Systems integration group, this group highlighted the expectation that trading would be handled by an automated system, in line with preferences set by participants. During the interview, participants highlighted the need to distinguish between features that would be required for these systems to operate and those required for them to scale. Interviewees also highlighted the distinction between interoperable and standardised, noting that interoperability of devices and data within the system would be required in order for them to function, whereas standardisation would be important for scaling.

In the Transactions and Markets group, seven defining characteristics were identified: prosumer-centric markets; active participation from end-users; the promotion of local generation and consumption; decentralised markets and generation assets; a market designed to capture value from flexibility and balancing; the ability to trade commodities i.e. energy or services; and for markets to be enabled by digitalisation and the integration of devices and communication. The role of prosumers in the market was an important theme during the interview, with some debate about the extent to which end-users could be expected to make 'active' decisions in order for the market to be considered as belonging to the P2P/TE/CSC class of models. There was general consensus that users would likely have the option to be active participants, by setting preferences for trading and optimisation of their energy use and generation, but much of the trading would be done on their behalf through an automated system. Interviewees also highlighted that many energy markets feature digitalisation, meaning that this characteristic does not necessarily distinguish these models from traditional energy systems, but it would be essential for P2P/TE/CSC markets to be feasible. Participants also felt it was important to highlight that decentralised energy markets might involve trading energy commodities other than electricity, such as heat or grid services and flexibility.

In the interview focusing on social and economic value, eight defining characteristics of P2P/TE/CSC models were identified: the promotion of local production and consumption of energy/electricity; the inclusion (implicitly or explicitly) of both monetary and non-monetary returns; a clear demonstration of the exercise of rights; open and equitable access through a bottom-up market design; markets simultaneously located in socially different spheres such as the home, within the community, and the wider market; inclusion of market-determined pricing, including dynamic pricing; systems that generate, modify and reinforce social values; and the involvement of a minimum of two social units (i.e. an energy-giver and energy-receiver). In addition to characteristics relating to local generation and consumption that were also noted by previous expert groups, interviewees in the social and economic value group highlighted the importance of social relationships within these models. Interviewees noted that these relationships can take place in different spheres, from the home, to the community, to the wider market. There was discussion about how, unlike traditional energy systems, P2P/TE/CSC models have the potential to create new forms of social value and relations, but also to change exisiting social dynamics. Financial incentives for participation were mentioned, as was the need for market-driven pricing, but the importance of nonfinancial aspects was also strongly endorsed by interviewees as a factor that distinguishes these models from traditional energy systems. The rights of participants (e.g. the right to ownership, right to consumption) also featured in the discussions. In order for these social, economic, and legal features to be realised, interviewees noted that there needs to be at least two uniquely identifiable agents to engage in an energy transaction. This could be a household or a device. Interviewees also

	Support integration of local generation	5.50
tion	Decentralised	5.10
tegra	Use local distribution networks	5.00
ns in	Smart energy interface and control mechanism	4.50
sten	Participants need smart meters	4.50
er s)	Support integration of renewable energy	4.20
Pow	Promote energy self-consumption	4.10
	Systems operate autonomously	3.10
	They include Advanced Metering Infrastructure ('smart meters') capable of measuring flows (imports and exports) across the meter at different time frames depending on market requirements The data is in an interoperable format interpretable by all parts of the P2P/TE/CSC system. Data can be shared with other actors in the system. The components of the P2P/CSC/TE system (generators, storage, DSM technology, etc.) have APIs (application programming interface) that allow if a the overhead of the and emitted commande Advancements.	8.71 7,43 5.71
ita	Distributed infrastructure	4.57
sb br	Consumer interfaces exist that allow participants to express trading preferences for different energy attributes	4.43
CT ai	Traceable identity of machines	4.29
-	Independent agents must exist at device level. These agents act on behalf of the user, based on user preferences	4.00
	The data can be shared in such a way as to comply with local data protection laws.	3.71
	The ICT has the capacity to deal with current and future frequency, diversity and volume of data.	2.14
	Prosumer-centric markets	4.63
rkets	Active participation from end-users	4.58
l ma	Promote local generation/consumption	4.37
s anc	Decentralisation	4.21
tion	Capture value from flexibility and balancing	3.53
ansad	Ability to trade commodity i.e. energy, or services	3.37
Ţ	Enabled by digitalisation and integration of devices and communication	2.79
	Promotes local production and consumption	6.18
lue	Both monetary and non-monetary returns are (implicitly and explicitly) part of the models	5.36
ic va	Clear demonstration of the exercise of rights (e.g. rights to consumption, rights of ownership)	5.00
non	Open and equitable access to P2P/TE/CSC models (bottom-up market design)	4.64
d ecc	Simultaneously located in home, community and market spheres of economy	4.36
al an	Market determined pricing (including dynamic pricing)	3.82
Socia	Generate/modify/reinforce social values	3.45
	Minimum of two social units (an energy-giver and energy-receiver) are involved	3.18
5	Exchanging energy through trading and/or sharing	4.36
latio	Active end-user involvement including through automation	4.00
regu	Enabled by digital infrastructures including platforms that can deliver multiple services	3.55
and	Legal recognition (obligations and rights) of prosumers	3.55
olicy	Bottom-up governance model reflecting decentralisation	3.09
ď	Promotes investment in local assets	2.45
	c	2 4 6 8 10 Weighted average

Fig. 3. Top defining characteristics by sub-task expert group.

pointed to the importance of context, noting that in the Global South, energy access and fuel poverty is a much more important aim of these systems than decarbonisation or grid flexibility.

The final group, policy and regulation, identified six defining characteristics of P2P/TE/CSC models: energy exchanged through trading and/or sharing; active end-user participation, including through automation; platforms enabled by digital infrastructure, including platforms that can deliver multiple services (e.g. trading of energy or flexibility); legal recognition (i.e. obligations and rights) of prosumers; a bottom-up governance model reflecting decentralisation; and the promotion of investment in local assets (e.g. storage). Whilst many of the characteristics listed by this group overlap with previous expert groups, their discussion of them focused on the regulatory and legal perspective. Interviewees felt it was important to make a distinction between sharing and trading electricity and other energy services, noting that, although energy sharing is mentioned many times in the EU Directive RED II, it is not properly defined. Interviewees also noted that the closest equivalent to P2P in the current market is sleeving arrangements. There was also discussion of automation as a way of reducing the effort required to engage with the system. Interviewees felt that being an 'active' participant can include a one-off setting of preferences, which are then automated. Much emphasis was also placed on the rights of prosumers and consumers within the system: interviewees stated that from a regulatory standpoint, P2P/TE/CSC models involve the recognition of the principle of prosumer responsibility i.e. the prosumer will be legally responsible for his or her actions or inactions from a supply, distribution and generation perspective. From a settlement point of view, these models all assume that the prosumer will ultimately bear legal responsibility for his or her balance position. Overall, interviewees felt that, from a regulatory perspective, there are strong distinctions between these models and it is difficult to define them as a class.

To summarise, the findings across all expert interviews indicate common themes of automation as necessary to enable consumer use of these models, as well as the importance of values such as locality and community. Expert discussions also highlight the inclusion of all types of (geographic) scale of models, even solely virtual P2P schemes, which encompass both trading that aligns generation and demand in space and time and trading that only aligns generation and demand in time. However, there were also some contradictions between characteristics identified in each group interview. These common themes and contradictions are discussed further in Section 5 and triangulated with findings from the literature review.

4. Differentiation of models: Literature review findings

The previous section of the paper outlined the characteristics that define the class of models described as P2P/TE/CSC according to experts in the field. The second part of this paper focuses on identifying the characteristics which set P2P, TE and CSC models apart from one another and those that they have in common.

4.1. Qualitative content analysis

During the search process as outlined in Section 2.2, it was found that many of the papers that met our search criteria used several other terms interchangeably with P2P/TE/CSC. As the aim of this paper is to reflect how researchers and practitioners refer to these systems in practice, it was included these additional terms in the data extraction process. Given that these terms are co-evolving and that there is no globally accepted definition, it was deemed important to capture the diversity of terms used in this conceptual space.

Once the additional samples of papers had been identified, definitions, characteristics, and terms were extracted from the literature. Statements describing features that the system is, has, enables, or does were coded as *Characteristics*. Labels given to the system described were coded as *Terms*.

This data was coded in NVivo, a qualitative data analysis software, using a combination of deductive and inductive qualitative content analysis [17]. Two coders each independently coded half of the sample using inductive qualitative content analysis to identify unique, distinct characteristics. Results were then refined through discussion between coders. Each coder then used the agreed list of terms and characteristics to deductively check each other's half of the sample until agreement was reached between coders. Characteristics were initially coded as specifically as possible, and later grouped into broader themes for ease of presentation in the paper. Each identified characteristic was also coded against the term used in the document describing it.

In interpreting these results, it is important to note that a characteristic being coded against a term only means that it was observed at least once in our snapshot of literature.

4.2. Analysis of terms

After data cleaning, a final list of 95 characteristics and 36 terms were identified that were associated with the terms P2P, TE and CSC. Please see Fig. 4 and supplementary material for a full list of characteristics and terms.

In order to understand which broader terms are being used in the literature to refer to P2P, TE and CSC models and how similar they are to one another with regards to the number of characteristics they share, a similarity matrix was calculated using the similarity index (SI). The similarity matrix can be seen in Fig. 4.

Each term was treated as a vector with a series of 1s and 0s indicating whether or not each characteristic was present for that term (without an indication of how many times). Eq. (1) outlines the similarity index calculated. The resulting matrix indicating the similarity of terms can be seen in Fig. 4.

$$SI = \frac{|Term_H \cap Term_V|}{|Term_H|} \tag{1}$$

The value of SI in each cell of the matrix indicates how many characteristics the vertical terms $Term_V$ share with the horizontal terms $Term_H$. For example, the term *P2P energy sharing* has a total of 25 characteristics of which it shares 20 characteristics with the term *P2P energy trading*. This would mean the SI between *P2P energy sharing* and *P2P energy trading* is 0.8.

The terms *P2P energy trading* and *transactive energy* were the terms most frequently appearing in the definitions analysed and are often used interchangeably, sharing many characteristics. The similarity index between TE and P2P is higher than vice versa. This indicates that *transactive energy* is defined more broadly, while *P2P energy trading* provides more detailed characteristics. Alongside P2P energy trading several other terms have been used describing the same or a similar concept including *P2P energy sharing*, *P2P electricity supply* and *P2P energy market*. The terms *Transactive control* and *Transactive energy systems* were often used synonymous with TE or would describe sub-concepts of it. Other terms such as *Renewable Energy Community* and *Citizen Energy Community* were predominantly used in non-academic literature. Both of these terms are defined in the revised EU Renewable Energy Directive (RED II) [9]. Similarly, the term *Collective self-consumption* can be attributed to the EU regulatory framework [18].

In order to reduce the noise caused by the constantly evolving nature of the field and establish the key prevailing concept, terms expressing the same or similar fundamental concept were merged together. Ideally speaking, terms should share 100% of their characteristics in order to be considered as the same concept and, therefore, merged into one. This would require an SI of 1. However, given this situation, there are likely to be characteristics associated with a specific context. For this reason, a more lenient threshold of 80% and merged terms that had an SI ≥ 0.8 were used. Terms which met this

	Behind the meter trading	Citizen energy community	Collective self consumption	Community based market	Community energy	Community microgrid	Connected community	Decentralised energy market	Decentralised P2P trading	District power models	Energy Community	Energy sharing	Federated power plant	Flexibility markets	Flexibility platform	Local collective use	Local energy community	Local energy market	P2P electricity trading	P2P energy market	P2P energy sharing	P2P energy trading	P2P energy trading network	P2P microgrid	P2P power trade	P2P supply	P2P trading of renewable energy	P2P transactive models	Peer-to-community platforms	Prosumer community group	Renewable energy community	Transactive control	Transactive energy	Transactive energy systems	Virtual Power Plant	Virtual self consumption	Number of characteristics
Behind the meter trading	1	0.08	0	0	0.09	0.08	0	0	0.18	0	0.1	0.11	0	0	0.33	0	0	0.1	0.07	0.08	0.08	0.05	0.07	0.29	0.1	0	0	0.1	0.15	0.11	0.06	0.25	0.04	0.1	0.13	0	3
Citizen energy community	0.67	1	0.31	0.33	0.65	0.35	0.33	0.33	0.46	0.5	0.67	0.22	0.6	0.25	0.5	0.33	0.8	0.38	0.28	0.23	0.32	0.2	0.29	0.43	0.2	0	0.17	0.2	0.54	0.56	0.94	0.25	0.18	0.1	0.5	1	24
Collective self consumption	0	0.21	1	0	0.22	0.23	0.11	0.17	0	0.83	0.14	0	0	0	0	0.67	0.4	0.14	0.14	0.12	0.12	0.14	0.29	0	0.2	0	0.17	0.2	0.31	0.11	0.13	0	0.08	0.1	0.13	0	16
Community based market	0	0.08	0	1	0.09	0.15	0	0.17	0.09	0	0.05	0	0.2	0.25	0	0	0	0.1	0.07	0.04	0.12	0.07	0	0	0.1	0	0	0.1	0	0.22	0.13	0	0.06	0	0.13	0	6
Community energy	0.67	0.63	0.31	0.33	1	0.39	0.33	0.25	0.36	0.5	0.62	0.33	0.6	0.25	0.33	0.17	0.8	0.33	0.28	0.27	0.32	0.22	0.5	0.29	0.1	0.33	0.17	0.2	0.46	0.56	0.69	0.5	0.22	0.3	0.13	0	23
Community microgrid	0.67	0.38	0.38	0.67	0.44	1	0.33	0.25	0.27	0.67	0.33	0.22	0.4	0.5	0.33	0.5	0.6	0.52	0.35	0.31	0.36	0.31	0.36	0.14	0.3	0.33	0	0.2	0.46	0.56	0.38	0.25	0.28	0.4	0.5	0	26
Connected community	0	0.13	0.06	0	0.13	0.12	1	0.17	0.09	0.17	0.14	0.56	0.4	0.25	0.17	0	0.4	0.24	0.17	0.15	0.24	0.15	0.36	0.14	0.2	0	0.17	0.1	0.23	0.22	0.13	0.25	0.16	0.1	0	0	9
Decentralised energy market	0	0.17	0.13	0.33	0.13	0.12	0.22	1	0.27	0.17	0.1	0.22	0.6	0	0.17	0	0.2	0.24	0.24	0.31	0.24	0.17	0.29	0.14	0.2	0	0	0.3	0.15	0.22	0.25	0.25	0.2	0	0.25	0	12
Decentralised P2P trading	0.67	0.21	0	0.17	0.17	0.12	0.11	0.25	1	0	0.1	0.22	0.4	0.25	0.5	0	0	0.24	0.28	0.27	0.24	0.17	0.21	0.57	0.3	0.17	0.5	0.2	0.39	0.11	0.25	0.5	0.22	0.1	0.25	1	11
District power models	0	0.13	0.31	0	0.13	0.15	0.11	0.08	0	1	0.05	0.11	0.2	0	0	0.5	0.2	0.14	0.07	0.08	0.12	0.03	0.21	0	0	0	0	0	0.23	0.11	0.06	0	0.02	0	0	0	6
Energy Community	0.67	0.58	0.19	0.17	0.57	0.27	0.33	0.17	0.18	0.17	1	0.22	0.2	0.25	0.33	0.17	0.8	0.24	0.21	0.19	0.24	0.19	0.14	0.29	0.1	0.17	0.17	0.2	0.39	0.67	0.63	0.25	0.16	0.3	0.13	0	21
Energy sharing	0.33	0.08	0	0	0.13	0.08	0.56	0.17	0.18	0.17	0.1	1	0.4	0.25	0.17	0	0.2	0.24	0.21	0.23	0.24	0.15	0.36	0.43	0.1	0.17	0.33	0.1	0.31	0.11	0.06	0.25	0.16	0.1	0	0	9
Federated power plant	0	0.13	0	0.17	0.13	0.08	0.22	0.25	0.18	0.17	0.05	0.22	1	0.25	0.17	0	0.2	0.19	0.14	0.15	0.08	0.07	0.14	0	0	0	0	0.1	0.15	0.11	0.13	0	0.1	0.1	0.13	1	5
Flexibility markets	0	0.04	0	0.17	0.04	0.08	0.11	0	0.09	0	0.05	0.11	0.2	1	0.17	0.17	0	0.1	0.07	0.04	0	0.03	0.07	0	0.1	0	0	0	0	0	0.06	0	0.06	0.1	0.13	0	4
Flexibility platform	0.67	0.13	0	0	0.09	0.08	0.11	0.08	0.27	0	0.1	0.11	0.2	0.25	1	0	0	0.14	0.1	0.12	0.08	0.09	0.07	0.14	0.2	0	0.17	0.2	0.08	0.11	0.13	0.5	0.1	0.1	0.38	0	6
Local collective use	0	0.08	0.25	0	0.04	0.12	0	0	0	0.5	0.05	0	0	0.25	0	1	0	0.1	0.03	0.04	0	0.03	0.07	0	0	0.17	0	0	0	0	0.06	0	0	0	0	0	6
Local energy community	0	0.17	0.13	0	0.17	0.12	0.22	0.08	0	0.17	0.19	0.11	0.2	0	0	0	1	0.1	0.03	0.08	0.08	0.03	0.07	0	0	0	0	0	0.08	0.11	0.19	0	0.04	0	0	0	5
Local energy market	0.67	0.33	0.19	0.33	0.3	0.42	0.56	0.42	0.46	0.5	0.24	0.56	0.8	0.5	0.5	0.33	0.4	1	0.38	0.5	0.56	0.32	0.5	0.71	0.5	0.17	0	0.4	0.39	0.33	0.31	0.25	0.33	0.4	0.5	0	21
P2P electricity trading	0.67	0.33	0.25	0.33	0.35	0.39	0.56	0.58	0.73	0.33	0.29	0.67	0.8	0.5	0.5	0.17	0.2	0.52	1	0.54	0.48	0.44	0.64	0.71	0.5	0.5	0.33	0.7	0.69	0.44	0.31	0.5	0.49	0.5	0.5	1	29
P2P energy market	0.67	0.25	0.19	0.17	0.3	0.31	0.44	0.67	0.64	0.33	0.24	0.67	0.8	0.25	0.5	0.17	0.4	0.62	0.48	1	0.44	0.37	0.43	0.71	0.4	0.33	0.83	0.7	0.39	0.33	0.31	0.75	0.39	0.5	0.38	0	26
P2P energy sharing	0.67	0.33	0.19	0.5	0.35	0.35	0.67	0.5	0.55	0.5	0.29	0.67	0.4	0	0.33	0	0.4	0.67	0.41	0.42	1	0.34	0.57	0.71	0.5	0.33	0.5	0.4	0.54	0.44	0.44	0.5	0.41	0.4	0.38	0	25
P2P energy trading	1	0.5	0.5	0.67	0.57	0.69	1	0.83	0.91	0.33	0.52	1	0.8	0.5	0.83	0.33	0.4	0.91	0.9	0.85	0.8	1	0.86	0.86	0.9	1	0.83	0.9	0.69	0.89	0.56	1	0./1	0.7	0.88	0	59
P2P energy trading network	0.33	0.17	0.25	0	0.3	0.19	0.56	0.33	0.27	0.5	0.1	0.56	0.4	0.25	0.17	0.17	0.2	0.33	0.31	0.23	0.32	0.2	1	0.29	0.4	0	0.17	0.3	0.39	0.11	0.19	0.5	0.2	0.1	0.13	0	14
P2P microgrid	0.67	0.13	0.12	0.17	0.09	0.04	0.11	0.08	0.30	0	0.1	0.33	0	0.05	0.17	0	0	0.24	0.17	0.19	0.2	0.1	0.14	1 20	0.2	0.17	0.17	0.2	0.31	0.11	0.13	0.25	0.12	0.2	0.13		
P2P power trade	0.33	0.08	0.13	0.17	0.04	0.12	0.22	0.17	0.27	0	0.05	0.11	0	0.25	0.33	0.47	0	0.24	0.17	0.15	0.2	0.15	0.29	0.29	1	0	0.17	0.4	0.15	0.11	0.13	0.25	0.18	0.1	0.13		10
P2P supply	0	0	0.00	0	0.09	0.08	0.11	0	0.09	0	0.05	0.11	0	0	0.17	0.17	0	0.05	0.1	0.08	0.08	0.1	0.07	0.14	0.1	0.47	0.17	0.1	0.15	0	0.12	0.25	0.08	0.1	0		D
P2P trading of renewable energy	0.22	0.04	0.06	0.17	0.04	0.00	0.11	0.25	0.27	0	0.05	0.22	0.2	0	0.17	0	0	0.10	0.07	0.19	0.12	0.09	0.07	0.14	0.1	0.17	0.17	0.1	0.15	0.11	0.13	0.25	0.16	0.2	0.12		10
P2P transactive models	0.53	0.08	0.15	0.17	0.09	0.08	0.22	0.25	0.16	0.5	0.24	0.11	0.2	0	0.33	0	0.2	0.19	0.24	0.27	0.10	0.15	0.21	0.29	0.4	0.17	0.22	0.1	0.08	0.11	0.15	0.5	0.10	0.2	0.13		10
President community platforms	0.07	0.29	0.25	0.22	0.20	0.25	0.35	0.17	0.40	0.17	0.24	0.11	0.4	0	0.17	0	0.2	0.14	0.51	0.19	0.16	0.15	0.50	0.37	0.2	0.55	0.55	0.1	0.15	0.22	0.25	0.25	0.06	0.1	0.12		15
Prosumer community group	0.35	0.21	0.00	0.35	0.22	0.19	0.22	0.22	0.09	0.17	0.29	0.11	0.2	0.25	0.22	0.17	0.2	0.14	0.14	0.12	0.10	0.14	0.07	0.14	0.1	0	0.22	0.1	0.15	0.56	1	0.25	0.00	0.1	0.15		16
Transactive control	0.33	0.03	0.15	0.55	0.40	0.23	0.11	0.00	0.30	0.17	0.40	0.11	0.4	0.25	0.33	0.17	0.0	0.24	0.17	0.15	0.20	0.15	0.21	0.25	0.2	0	0.33	0.2	0.02	0.11	0.06	1	0.10	0.1	0.38		10
Transactive control	0.53	0.04	0.25	05	0.09	0.54	0.11	0.082	0.10	0.17	0.03	0.11	1	0.75	0.55	0	0.4	0.81	0.86	0.12	0.08	0.61	0.14	0.14	0.1	0.67	0.17	0.2	0.08	0.11	0.00	1	0.00	0.1	0.63	1	4 E 1
Transactive energy systems	0.33	0.04	0.06	0.5	0.40	0.15	0.11	0.85	0.00	0.17	0.30	0.11	0.2	0.75	0.05	0	0.4	0.19	0.17	0.10	0.16	0.12	0.07	0.30	0.5	0.07	0.00	0.0	0.08	0.11	0.06	0.25	0.18	1	0.05	0	10
Virtual Power Plant	0.33	0.04	0.06	0.17	0.13	0.15	0.11	0.17	0.18	0	0.05	0.11	0.2	0.25	0.1/	0	0	0.19	0.14	0.19	0.10	0.12	0.07	0.14	0.1	0.1/	0	0.2	0.00	0.11	0.00	0.23	0.10	0.1	1		10
Virtual self consumption	0.55	0.04	0.00	0.17	0.04	0.15	0	0.1/	0.00	0	0.00	0	0.2	0.23	0.5	0	0	0.19	0.03	0.12	0.12	0.12	0.07	0.14	0.1	0	0	0.1	0.08	0.11	0.15	0	0.02	0.1	0	1	• 0 1
• in coal soil consumption	0	0.04	0	0	0	0	0	0	0.03	0	0	0	0.2	0	0	0	0	0	0.00	0	0	0	<u> </u>	<u> </u>	0	0	0	0	0.00	<u> </u>	v	v	0.02	v	v	-	1

Fig. 4. Term similarity matrix.



Fig. 5. Terms merged based on number of shared characteristics with similarity index (SI) indicated in brackets.

condition could be considered sufficiently similar to be merged. Using this approach P2P, TE and CSC emerged as the three dominant terms with all other terms being either a sub-concept of one of the three terms or a separate unique concept. The merging of terms took place by only considering the number of characteristics shared to account for the interchangeable and inconsistent use of terms. This should not mean that the term or concept can be considered a standalone term in certain disciplines. The threshold value used influences the results of this analysis and is an important limitation to note.

A total of 26 terms were merged into the three dominant terms. There were 10 terms which did not have enough shared characteristics to be merged with P2P, TE, or CSC and were not considered in the further analysis (see Fig. 5).

An overview of the steps taken can be seen in Fig. 6.

4.3. Analysis of characteristics

In this section, the third research aim is addressed: to understand which characteristics are associated with P2P, TE, and CSC respectively, and where the similarities and differences between them lie.

After removing characteristics of the unique terms that were not shared with the merged terms, a total of 80 characteristics were identified across all terms. It is important to note that if a characteristic has not been coded as part of a particular term, this does not necessarily mean that this characteristic cannot be associated with the term, but



Fig. 6. Overview of quantitative content analysis and results.

rather that the characteristic was not observed in the literature reviewed that mentioned this term. This is an important limitation of the analysis which, by necessity of resource constraints and the immature state of the field, focused on a snapshot of literature.

Discussion of characteristics associated with the three main terms has been organised according to seven main themes. These themes were implemented to allow for clearer visualisations and identification of similarities and differences between terms across key dimensions of these models and have no order of importance. An overview of each characteristic and its associate reference can be viewed in Table A.9.

4.3.1. Market or system aims and incentives

Table 2 shows characteristics associated with each model under the theme of market or system aims and incentives; a coloured square indicates that at least one source in the literature sampled associated this characteristic with this model.

All three models shared the aim of *balancing of demand and supply*. For TE, this characteristic features prominently in the Gridwise Architecture Council definition: "TE refers to economic and control mechanisms that allow the dynamic balance of supply and demand across the entire electrical infrastructure, using value as a key operational parameter" [19]. This definition appears to have influenced other descriptions of TE as similar descriptions are found in [20–23]. P2P and CSC both share the additional goal of *reducing energy loss*.

Characteristics excluded by CSC, but shared by TE and P2P tend to relate to benefit the overall system (e.g. management of grid constraints; promote Renewable Energy Sources (RES) and DER generation; optimisation of energy behaviour to benefit system; aggregation of participant loads; flexibility trading) [e.g 2,3,20,23–40].

Incentives for participation in TE and P2P models include *economic incentives*. Notably, TE seems to frame economic value as an incentive,

Table 2

Characteristics explicitly \blacksquare and not explicitly \blacksquare associated with terms related to market or system aims and incentives.

Characteristic	P2P	TE	CSC
Balancing of demand and supply			
Self sufficiency			
Reduce energy loss			
Promote or include RES generation			
Promote or include DER generation			
Participants have greater control over preferences			
Optimisation of energy behaviour to benefit system			
Management of grid constraints			
Flexibility trading			
Energy democracy			
Economic incentives for participation			
Aggregation of participant energy loads			
Social benefits			_
Shared benefits across the community			
Identification of origin of energy supply			
Environmental benefits			
Community as focal point for engagement			

whereas P2P presents it as an outcome. For TE, economic incentives are often mentioned in connection to incentivising grid optimisation behaviour [19] and by descriptions of "demand-side parties optimis[ing] their behaviour in response to economic signals, to minimise overall energy cost" [37]. For P2P, economic incentives are mentioned both in terms of unspecified economic benefits and more specific benefits for prosumers such as earning revenues and reducing electricity costs [41] or for system operators in terms of lowering operational costs [20].

Both P2P and TE also refer to *self-sufficiency*. For both models this is described as reduced dependence on the grid [41,42], whereas

Table 3

Characteristics explicitly \blacksquare and not explicitly \blacksquare associated with terms related to participant type and scale.

Characteristic	P2P	TE	CSC
Consumers with photovoltaic (PV) solar systems small-scale participants			
Prosumers			
Equal in size market participants			
Diverse types of energy generators and consumers			
Individuals or at communities			
Groups of participants			
Commercial power consumers			

descriptions of P2P also explicitly mention reduced reliance on a central supplier and increased self-determination and autarky for participants [43]. This is perhaps reflective of the more individualistic and decentralised nature of P2P, discussed in Section 5.2.1.

Both P2P and TE also focus on giving participants more power and control over their energy consumption through *energy democracy* and *greater control over preferences*, empowering individuals to participate in energy markets. P2P takes this a step further, mentioning that P2P transactions could be traced back to their origin [44]. This associates P2P with a move away from purely financial or system-focused benefits and into alternative value propositions, such as social and environmental benefits. This is less explicit for TE: the Gridwise Architecture Council [19] mentions "a wide array of societal benefits" but does not state the nature of these benefits or who they are delivered for. Similarly, [23] mention that TE can bring "social dimensions" to the grid, but does not specify whether these are social benefits.

Characteristics unique to CSC involve shared benefits across the community and community as a focal point for engagement. For example, [45] describes community microgrids as aiming to "maximiz[e] the social welfare of the community" in a manner that ensures that the solution for each member is at a minimum not worse than an individual outcome. [46] also makes reference to "shared identity or desire for strong, self-reliant communities". This suggests that there is a more explicit focus on community-oriented, social value propositions for this model.

4.3.2. Participant type and scale

Table 3 shows that the presence of *consumers with PV solar systems* is common across all models. This is the only type of participant specifically mentioned for CSC models; however, unlike P2P and TE, descriptions of CSC focus on shared access. For example, [47] describes CSC as allowing "a group of actors [to consume] electricity from a shared PV system" and [48] characterises building-scale CSC as allowing "several electricity consumers located in one single building [to] benefit from the electricity produced by a PV system installed on the same building".

Papers discussing TE and P2P both mention the ability for diverse types of energy generators and consumers to participate, including the broad category of prosumers. For P2P, definitions categorised under this characteristic vary: [49] summarises participants in P2P as consisting of "a variety of players, with equal access to a common resource and goal of sharing it through a wealth of cooperating infrastructures", whereas [25] emphasises the "diversity of the generation [...] and demand profiles of different customers". [32] notes that a "peer can be a single person or an authority" whilst [20] mentions the ability for P2P to be "employed at higher levels for energy trading between microgrids, energy communities, or VPPs [Virtual Power Plants]". These broad, and at times contradictory, definitions indicate that P2P is used to refer to a wide range of systems. Although diverse types of energy generators and consumers is also coded for TE, the focus here is more on the heterogeneity of participants, in terms of their type [24,50] and size [51].

Some sources for P2P highlight that participants should be *small-scale* and that trading takes place between participants that are *equal*

in size [e.g. 31,32,52–54]. In line with the market aim of increased self-sufficiency, several sources for P2P also highlight that participants can be entities previously excluded from the energy market, now able to trade energy without reliance on the retail market, centralised suppliers, or large companies [20,32,41,52,55,56]. Although *smallscale participants* and *equal in size* are also coded for TE, this is only mentioned by [32], which describes TE as providing an "opportunity for small and mid-sized energy consumers or producers to exchange energy".

P2P is the only model to mention that commercial consumers can also be included [57]. This indicates that P2P places greater emphasis on the decentralised nature of transactions than necessarily on the type of participants involved. In this vein, [38] mentions that groups of participants can take part in P2P and contract as a coalition with a retail supplier.

4.3.3. Participant interactions

As shown in Table 4, texts discussing CSC did not specify how participants within the system might interact with one another. Texts describing TE and P2P mention both the possibility of *autonomous communication between participants* [29,39]. However, both models also highlight the possibility for *active end user participation*. For TE, this is largely characterised in terms of allowing control over their energy transactions and a means of procuring value for the grid [20,22]. For P2P, the theme of active end user participation is also described in terms of active negotiation of energy procurement [58,59], but also highlights consumer empowerment in the form of "greater freedom to engage in transactions with partners of their choice" [54] as well as having "more choice over where they purchase energy from" [60]. This highlights the importance of non-financial value for P2P.

Under TE and P2P, market participants are considered to be *independent and self-interested decision makers* [41,44,61]. Notably, [62] refer to actors in both P2P and TE as behaving in a rational and individualistic manner in order to maximise their own profit. Conversely, [49] mentions the possibility of P2P actors having *shared energy trading goals*: "One can see it as a variety of players, with equal access to a common resource and goal of sharing it through a wealth of cooperating infrastructures, which is opposite to the traditional economic principle represented by players with individual goals". This particular definition might be more aligned with what might typically be considered to be a CSC project, demonstrating the wide range of projects that have been labelled as P2P and the inconsistencies between the uses of these labels.

P2P also places emphasis on the freedom of trading and participation. [63], describes P2P as "the most challenging market design arrangement; consumers and prosumers have the possibility to freely trade with a central part or, in its purest form, between each other". Relatedly, [64] emphasise that participants should be able to be added or removed from a P2P network without adverse effects on network stability.

4.3.4. Market or system design

Characteristics related to the design of the market as seen in Table 5 are limited to P2P and TE models. For both P2P and TE market coordination mechanisms range from centralised to distributed to decentralised. *Centralised coordination and management* in TE is only mentioned once and is associated with P2P energy sharing [30]. Generally speaking, TE puts a stronger focus on decentralised and distributed management for better privacy [65] and system scalability [39]. In P2P the key aim of a decentralised or distributed system is to avoid a central intermediary party [20,40,61,66,67].

However, [28] highlights that P2P energy trading might "rely on some degree of centralised communication and control, rather than being strictly based on P2P negotiation and autonomous decision making". A similar view is supported by [35] in the case of VPPs. [68] considers both possibilities, using centralised controller to manage a market or relying on decentralised technologies to avoid a centralised

Table 4

Characteristics explicitly	and not	explicitly	associated	with to	erms related	to pa	articipant	interactions
Charactoristic					00	D.	TE	CSC

Characteristic	PZP	IE	CSC
Participants are independent and self interested decision makers			
Autonomous communication between participants			
Active end-user participation			
Participants trade freely with one another			
Participants have shared energy trading goals			
Participants can be added or removed without affecting network			

Table 5

Characteristics explicitly \blacksquare and not explicitly \blacksquare associated with terms related to market or system design.



Table 6

Characteristics explicitly \blacksquare and not explicitly \blacksquare associated with terms related to market or system scale.

Characteristic	P2P	TE	CSC
Local energy trading or market (geographic region)			
Bound by LV-distribution network			
Virtual energy trading (no geographical restrictions)			
Operate in grid-connected mode			
Bound by local community			
Bound by distribution network			
Operates on various levels of the grid			
Operates on public electricity grid			
Operates on private electricity grid			
Operates in island mode			
Connected at distribution grid level			
Bound by building or block			

system is bound by the topology of the grid or locality of the community and to what extent. A frequent mention were the two opposing characteristics that describe whether a system is designed as a *local energy trading or market (geographic region)* or through *virtual energy trading (no geographical restrictions)*. Both P2P and TE explicitly consider both forms of market design. As highlighted in the context of TE: "it can be applied within a localised area, e.g., microgrid, or be utilised to manage the whole power system" [77]. Specifically the balancing of energy can be conducted "virtually" to benefit participants of these markets [75]. [47] describe this form of virtual energy trading as "virtual self-consumption, where generation and consumption of PV happens at the same time but in differing locations".

For TE, it is highlighted that the system can *operate across various levels of the grid* [19,21,24,78,79], starting at the local prosumer level and going up to the whole energy system, including the wholesale market. The levels of market operation strongly correlate with the participant types and scales involved in the market as described in Section 4.3.2. In summary, TE is less restricted by a specific level of the electricity grid compared to P2P or CSC.

CSC is the only model that explicitly limits the scale of the market operation for example by a building or block [48]. CSC also specifies that the sharing of electricity can be conducted on a *private or public electricity grid* [47,80]. Whether a market runs on a public or private electricity grid can impact the governance model of the market and ultimately the type of concept in use.

All three terms consider a certain degree of locality by restricting the market or trading place to be *bound by the LV-distribution network*. Other characteristics expressing a similar concept mentioned for P2P and TE included the market being *bound by a local community* [34,40] or *bound by the distribution network* [22,65]. Finally, *islanded operation* rather than grid-connected operation was considered in the case of P2P [64]. However, this was mentioned in association with the concept of microgrids [44,49,66], which was merged with P2P during the analysis of terms in Section 4.2.

In general it can be said that the concept of locality plays an important role in all three terms, not least to integrate DER efficiently into the energy system. However, boundaries of locality are case-dependent and can be considered in the context of a geographical region, the topology of the grid, or physical structure.

party. Centralised market management is associated with roles such as "community manager" [64], "licensed suppliers" [68] or "centralised controller" [30]. It can be observed that the more decentralised a market becomes, the stronger the focus on autonomous trading is in the case of P2P [1,67,69] and *autonomous grid control* in the case of TE [27,39,42].

The aspect of trading plays an important role in P2P and TE highlighted by characteristics such as *market rules and mechanisms* with a *competitive market structure and pricing* using e.g. "negotiations" [2] or "optimisation" [42]. Further, a market can incentivise different types of participant behaviour: "The members may work either in a collaborative or a competitive manner" [64]. P2P and TE make use of *market-based price signals*, to react to grid constraints and balance demand and supply of energy [2,19,20,35,38,42,52].

Pricing mechanisms used by both include *dynamic pricing* [20,41, 43,70], while P2P explicitly mentions *time-of-use tariffs* [71] and *nodal and local grid pricing* as a subcategory [72]. As outlined by [41]: "As an independent decision maker, a prosumer may intend to sell his or her surplus energy at different rates to different buyers within the network", highlighting the role of diverse pricing schemes in P2P to incentivise certain load behaviour or include consumer preferences.

P2P can operate as a *sub market within a traditional market* [73] and requires the *grid as the supplier of last resort.* While [74] mention that P2P could effectively replace the current role of the energy supplier, [75] state that P2P could contribute to a better balance of demand and supply alongside the services delivered by the supplier. Demand not satisfied by own generation could be met by the supplier [38,62]. This is in line with the characteristics mentioned under Section 4.3.1 describing the aims and incentives of the market or system.

Finally, TE markets have a stronger focus on *coordination and management of smart devices*, mainly customer devices [24,39,76], while P2P, being derived from the *concept of sharing economy* [25,75] focuses more on the prosumers and the relationships between them [1].

4.3.5. Market or system scale

Table 6 shows characteristics related to market and system scale. The characteristics mainly focus on specifying whether a market or

Table 7

Characteristics explicitly ■ and not explicitly ■ associated with terms related to market or system transactions and settlement.

Characteristic	P2P	TE	CSC
Real-time trading			
DLT for energy transaction and settlement			
Automatic execution and settlement			
Use of virtual or real currencies			
Trading of renewable energy			
Physically delivered energy			
Multi-directional exchange of energy and value			
Manual energy trading			
Frequent transactions			
Energy transactions without intermediary			
Energy transactions through intermediary			
Contracting for energy services offered and delivered			
Bi-directional energy flows			
Transactions of reactive power			
Transactions of real power			
Trading of different classes of energy			
Multilateral bidding transactions			
Forward energy transactions			
Deferred energy transactions			

4.3.6. Market transactions and settlement

Table 7 shows characteristics related to market transactions and settlements. Characteristics are predominantly represented by P2P and TE models. All three market models mention *automatic execution and settlement* of transactions as a key characteristic, while P2P and CSC also consider *manual energy trading*. In the words of [35], P2P "enables prosumers to buy and sell energy directly - manually or via automation - with a high degree of autonomy". In both concepts, the degree of automation can vary [35,81].

To conduct transactions in the market, all three models mention *DLT for energy trading and management* [26,35,81,82]. It is mainly used to establish trust between the parties involved and allow for a more decentralised and automated management of the market, e.g. through the use of smart contracts [40,44]. Amongst others, the use of this technology can also determine whether energy transactions happen *with or without an intermediary*. Avoiding intermediaries in some cases refers to making central electricity suppliers redundant [38,83,84] while in other instances it refers to specifically the removal of any "centralised supervisor" or mediator [49,64]. In these cases, the presence of bi-lateral or *multilateral bidding transactions* is necessary [41,67,71,85].

When intermediaries are considered, it is often in the form of a "central marketplace" [86,87] or an "aggregator" [18,29] that facilitates transactions between parties. The integration of a centralised third party is frequently mentioned in the context of the RED II definition [18,29,82].

A shared aim across all models, regardless of whether they conduct transactions with or without an intermediary is to avoid or reduce the dominance of and dependence on the traditional utility company when exchanging energy [56].

P2P and TE were also associated with *trading of renewable energy*, with TE further segregating the type of energy traded. In TE, both *real and reactive power* can be transacted [34]. P2P puts a stronger focus on participants' preferences and includes *trading of different classes of energy* [49,58,88]. Papers describe how participants can trade different classes of energy, with attributes going beyond financial considerations [44]. For example, those interested in locality could select the provenance of their electricity [71], environmentally conscious prosumers can buy renewable energy, and philanthropic prosumers could supply subsidised energy to low-income households [28].

For both terms, the *physical delivery of energy* plays a key role [54, 70] including through *bi-directional energy flows* [24,62,89]. As described by [54] the physical delivery of energy "impose[s] physical limits on the electricity trade". Market designs that consider the physical flow of energy would therefore have a stronger focus on locality

Renewable and Sustainable Energy Reviews 202 (2024) 114672

Table 8

Characteristics explicitly \blacksquare and not explicitly \blacksquare associated with terms related to governance.

Characteristic	P2P	TE	CSC
Shared ownership of generation assets			
Non-traditional business model			
Assets owned by customers and third parties			
Participants not selling energy professionally			
Owners have control over devices			
Extension of existing business models			
Equal responsibility and rights for all participants			
Community operates as legal entity			

as discussed in Section 4.3.5. Such market designs would ultimately be linked with the *multi-directional exchange of energy and value* [27,68,81, 90].

The time scale of the transactions was mainly highlighted by TE, which allows for *forward and deferred energy transactions* [26,70] meaning transactions can be settled prior or after their occurrence. All three models offer *real-time trading* in particular to "facilitate a sustainable and reliable generation and consumption of energy within the community" [91] and respond to short term needs of the grid [34,39, 59].

4.3.7. Governance and policy

Table 8 shows that the only characteristic of governance unique to TE is the explicit mention of *owners having control of their own devices* [39].

Another characteristic of TE also shared by P2P includes *assets owned by customers and third parties*. [82] highlights that DER assets are individually owned and installed behind-the-meter in P2P networks, and [21] notes that TE systems extend to all levels of the grid, including customer-owned and third-party assets.

The literature also discusses the disruptive nature of P2P and CSC as a non-traditional business model, in terms of allowing smaller suppliers to compete with large, traditional suppliers [61]. [74] goes so far as to say that "theoretically, P2P-supply could make the role of the traditional Supplier, i.e. a centralised profit-driven market party, redundant in the long term". By contrast, [54] describes P2P as an extension of existing business models, with the platform being operated by existing energy market players. This highlights a gap between the radical ambition of P2P and its practical implementation. Similarly, [92] describes TE as "liberating the data that will transform energy markets, and rewarding consumers with better energy services that come from utilising the existing grid and adding new infrastructure where it is needed most". Whilst texts describing P2P speak of the potential redundancy of traditional suppliers, representing a shift in power away from traditional entities and towards individuals, those describing TE appear to view this model as complementing existing markets rather than replacing them. This generally aligns with the characteristics discussed in Section 4.3.4.

P2P also mentions *shared ownership of generation assets*, a characteristic shared with CSC. [49] describes how, in P2P, in addition to individual trading, "a community can also be formed by prosumers who want to collaborate, or in terms of operational energy management". For CSC this is much more focal and participants are typically mentioned in terms of groups of consumers [47,48,80]. For CSC only, it is explicitly mentioned that the *community operates as a legal entity*.

Notably, P2P also highlights that all participants have *equal rights and responsibilities*, emphasising its focus on egalitarianism. Additionally, [54] highlights that a transaction can only be qualified as P2P when it is undertaken by two *non-professional actors*.

5. Discussion

5.1. Common themes

Several common themes emerged across all disciplinary perspectives in the expert group interviews and in the sample of literature reviewed.

5.1.1. Digitalisation and automation

The first common theme across all sub-task expert group interviews and the literature is that of automation. Most expert groups highlighted automation as an important characteristic of P2P/TE/CSC models to enable consumer participation. Similarly, in the literature, P2P and TE were associated with autonomous communication and grid control between participants.

Groups focused on the technical aspect of LEMs argued that automation is necessary since consumers cannot be expected to manually manage transactions in practice. This should not contradict the presence of active participation as highlighted by groups 3 and 5. Automation could reduce the effort required by consumers to be active e.g. with intervention only at the beginning to set parameters according to group 5 on policy and regulation. Different interpretations of end-user participation are also borne out in the literature. There is a distinction between 'participation' in the sense of allowing end-users to express their preferences without the need for continued engagement, and a more active form of participation involving behavioural change and active engagement with the energy system and energy trading platforms. Papers discussing TE tended to speak about end-user engagement in terms of the former, whereas papers discussing P2P tended to include both forms of participation.

On the broader topic of digitalisation, there was consensus across several expert groups that smart meters in particular are key to the functioning of P2P/TE/CSC models. Furthermore, groups 2 and 4 felt it essential that data collected should be in an interoperable format so that it can be shared with actors across the market and aggregated. In the literature, smart IT platforms were mentioned in papers discussing P2P and TE. For all three models, the use of DLTs for energy transactions and management was mentioned in the literature, while not specifically discussed in the expert interviews.

5.1.2. Values

Almost all expert groups agreed that P2P/TE/CSC models are characterised by their promotion and support of the local generation and consumption of energy or local assets. Similarly, in the literature, papers discussing P2P and TE both mentioned the promotion or inclusion of renewable and DER assets. Energy trading at the local geographic region is mentioned in the literature for all models. Locality is therefore a crucial value driving these systems.

According to expert interviews, actors in the system should be able to trade and/or share energy and related services. This can be in cash, in kind or in intangible payment. The concept of trading different energy classes was also mentioned in the literature on P2P. Group 3 on market design raised the point that P2P/TE/CSC systems can include a diverse resource portfolio, e.g. markets including power, heating, cooling, transport (via electric vehicles) and both thermal and electrical storage. Similarly, literature on P2P and TE mentioned the possibility for diverse types of energy generators or consumers to be included in the system.

As for consumers' motives for participating, in the expert interviews, there was consensus between group 3 on market design and 4 on social and economic value that both monetary and non-monetary values drive P2P/TE/CSC systems. These are framed as either individual or community-based (i.e. shared) benefits. This is broadly aligned with the literature reviewed, although these incentives appear to differ for P2P, TE, and CSC models. Economic incentives for participation were mentioned in papers discussing P2P and TE, whereas social benefits for individuals were mostly associated with P2P, and shared benefits across communities were a core feature of CSC.

5.1.3. Scale and governance

Several expert groups pointed to the fact that in most P2P/TE/CSC systems, energy is not being traded in a 'physical' sense as the origin of the electricity cannot be tracked once it is injected onto the grid, rendering the exercise of trading as depending on what can be netted. Therefore, models centred around the virtual trading of energy and distributed networks are also considered to be P2P/TE/CSC according to groups 1 on power systems and 3 on markets and transactions. In contrast, the reviewed literature highlighted that the trading of energy could be geographically restricted or traded virtually, whereby the term "physically" as used by the interviewed experts could refer to locally restricted markets.

Group 5 on policy and regulation also highlighted that there should be no size requirement. Many of the current pilots are being carried out within a microgrid, private wire or behind-the-meter setting. These should also be included under the class of models encompassing P2P/TE/CSC. This is strongly supported by the characteristics associated with CSC in the literature, where energy trading can be restricted to one building or block.

Furthermore, there was consensus across several expert groups that in order for these business models to become viable, they must be able to use the local or public distribution network. Pilots should be designed to function at a larger scale, in order to be integrated into the main energy grid. The latter implies that individuals or entities involved in pilots would have to offer services and take on obligations traditionally applicable to a licensed utility, such as balancing responsibilities, customer billing, and upholding data privacy rules- or partner up with a utility to do this. In the literature this feature was associated with the term TE, which focuses on providing grid services and engaging in energy trading across various levels of the grid. The consideration of grid services in the market activities played a key role in the market design characteristics for P2P and TE.

The legal recognition of trading entities, such as energy prosumers and communities, is therefore crucial. There needs to be a clear demonstration of the rights (e.g. right of ownership, right to consumption) and obligations of these new market entrants according to expert groups 4 and 5. This will require the setting up of verification processes and adaptation of the licensing framework. Government policies to encourage flexible behaviour by consumers will also be necessary as highlighted by group 5 on policy and regulation. While this was also described in the characteristics identified as part of the literature review, detailed information on verification processes and licensing frameworks is yet to be established. Nevertheless, CSC in particular put a strong focus on the shared ownership of generation assets and the operation as a legal entity.

When it came to market governance, in the interviews, group 5 on policy and regulation in particular but also groups 3 and 4 agreed that P2P/TE/CSC models should include a democratic and bottom-up governance model reflecting the decentralised nature of these systems. Stakeholders can input into and arrive at a consensus on how the market is designed or structured. There was a similar discussion in the literature on the values of energy democracy, particularly associated with P2P, as well as shared ownership of generation assets in P2P and CSC models. Papers discussing P2P additionally mentioned the possibility of participants being defined at either the individual or community level.

5.1.4. Market design

The topic of market design was discussed in all expert interviews. For group 3, the top priority was for these systems to be prosumercentric markets. Group 4 also highlighted the importance of open and equitable access to P2P/TE/CSC models, which must be designed in a user-centred manner, as well as include residential and commercial participants that are 'relatively small' in size (e.g. not a nuclear power station). Furthermore, markets need to be designed to allow for individual or communal ownership of distributed assets according to group 5.

Equally, communal assets were a key characteristic of CSC in the literature review. Social benefits and shared benefits were associated with P2P and CSC in the literature, respectively. While for CSC it was generally assumed that small-scale consumers with PV installations are involved, P2P and TE papers highlighted that the key targeted market participants should be equal in size in support of the group 4 experts' statement.

Group 4 on social and economic value placed particular importance on the personal dimension of market interactions. Trading needs to take place between two points, individual participants or a group of participants, which are identifiable and visible on the system. Additionally, participants in groups 1 and 3 suggested that transactions between participants could be direct, i.e. without any involvement of intermediaries. In contrast to this, in the literature, both trading with and without intermediaries was discussed for the concepts of P2P and TE.

Regarding market parameters, groups 1, 3 and 4 agreed that dynamic pricing is a key component of P2P/TE/CSC models. However, it was argued that dynamic pricing may only be one example of realtime trading, opening the way for other types of pricing mechanisms. The crucial point here is that the pricing system should be determined by the participants according to group 4. A local central market coordinator may also be necessary according to the experts. Group 4 raised the valuable point that non-market pricing should also be included, as observed in the Global South. Similarly, in the literature review various forms of real-time or dynamic pricing structures were discussed that could incentives a certain behaviour of the market participants. However, these flexible pricing strategies were mainly associated with the concepts of P2P and TE. The trading aspect was not discussed in detail in literature with regard to CSC, which might support the comment made by group 4 about non-market pricing-focused concepts.

In summary, there was broad agreement between the defining characteristics of P2P, TE and CSC as a whole identified in the expert interviews and in the literature review, albeit with different emphases given by particular models and disciplinary perspectives. The next section focuses on results obtained from the literature review discussing differences identified between P2P, TE, and CSC to establish an overview of the three models individually.

5.2. Comparison between models

Given the results and insights gained from Sections 3 and 4, the following sections aim to describe the terms P2P, TE and CSC. Rather than providing rigid definitions, the key characteristics identified in the expert interview and literature reviews are drawn upon, and key features or present boundary conditions that enable differentiation between the term are described.

5.2.1. P2P energy trading

Peer-to-peer energy trading is the most richly described term in the literature, with the greatest number of characteristics. Consequently, the projects described as P2P have the broadest range of characteristics. In terms of the system aims and incentives, P2P encompasses systemorientated benefits as well as social and environmental benefits, and increased control over preferences and self-sufficiency for participants. However, it does not mention community-focused aims. P2P can encompass a diverse range of participants, from individuals to groups to commercial consumers. Also emphasised is that participants are smallscale, non-professional, and equal in size. P2P is also distinguished from the other two models in that it places emphasis on the freedom of participants to come and go without affecting the network, and that participants are able to act as self-interested decision makers or with shared trading goals. Similarly, assets can be owned by individuals or shared. P2P can be operated in a central, distributed or decentralised way. Transactions can take place with or without an intermediary. The trading aspect is key to P2P with a particular focus on competitive

markets. They can operate within a sub-market alongside traditional markets i.e. fall under the operation responsibility of distribution system operators. P2P has been mentioned to be bound by a geographic area or consist of virtual energy trading. P2P can be an extension of existing models as well as a disruptive business model, and places emphasis on all peers having equal rights and responsibilities.

Based on this analysis, P2P is defined according to the following characteristics: it is a sub-market that can operate alongside traditional energy markets. Individuals can trade energy within a community, which can be bound locally or encompass virtual trading across a large geographical region. Participants can be heterogeneous in type but are typically small-scale and equal in size. Although P2P markets are set up to encourage competitive behaviour, with economic incentives and prosumers having individual trading preferences and goals, the overall market aim generally pertains to social, environmental, and energy democratisation benefits.

5.2.2. Transactive energy

Most of the aims and incentives associated with TE pertain to system benefits and optimising the integration of RES and DER generation. Benefits for participants are associated with self-sufficiency, democratisation of energy, control over preferences, and economic incentives. Participants act as individual, self-interested decision makers, again pointing towards a more individualistic incentive structure, albeit constrained by the grid. This is also reflected in the ownership of assets by consumers and third parties and in that owners have control over their own devices. In TE, participants can communicate autonomously, or be active participants. Similarly to P2P, TE highlights that participants can be diverse but should be small-scale and equal in size. TE markets can be operated in a central, decentralised or distributed way. They are dependent on an intelligent platform and use different forms of pricing mechanisms to trade energy. Both trading that aligns generation and demand in space and time and trading that only aligns generation and demand in time can be traded across various levels of the grid. TE also emphasises the non-traditional business model.

TE is therefore characterised as a non-traditional business model that allows energy end-users to have greater control over their energy trading preferences. It typically provides economic incentives for participants to trade energy in a manner that supports electricity grid balancing. The system can operate across various levels of the electricity grid. Similar to P2P, participants are typically small-scale and equal in size. Different types of energy can be traded. The primary value of TE tends to pertain to systems such as grid stability and reliability while supporting the increasing installation of DER and RES.

5.2.3. Collective/community self-consumption

CSC is the only model to put explicit focus on community-oriented aims and the sharing of benefits across the community. This is reflected in the shared ownership of generation assets. It also mentions benefits such as balancing demand and supply and reducing energy loss, but does not explicitly mention the system and economic aims and incentives highlighted by P2P and TE. The only types of participants explicitly mentioned are consumers with PV. Characteristics describing the market are limited. CSC usually operates within specific topological restrictive areas, i.e. the LV grid or a building or block. Energy trading can be conducted manually or through automatic execution and settlement. As CSC is the only model with a legal definition, it is also

Table A.9

Overview of characteristics and associated references.

Characteristic type	P2P	ТЕ	CSC
Market or system sime and incentives			
Market or system aims and incentives		-	
Balancing of demand and supply	[3,28,43,52,73,75]	[10,19–24,34,42,57,65,90,93–101]	[81]
Self-sufficiency	[41,43,74]	[42]	F013
Reduce energy loss		F0 4 00 10 41	[81]
Promote or include RES generation			
Promote or include DER generation		[20,24,34,39,42,65,104]	
Participants have greater control over preferences	[24,28,38,41,44,49,58,60,61,69,71,72,84,88, 107]	[20,28,30]	
Optimisation of energy behaviour to benefit system	[30,40,64,87]	[24,30,37,90,92,108]	
Management of grid constraints	[3,20,25,28,30,31,44,71,73,109]	[2,10,20,23,24,26-28,34,65,93,94,110]	
Flexibility trading	[3,25,31,72,88,111]	[24,70,79]	
Energy democracy	[54,84]	[39,104]	
Economic incentive for participation	[20,24,25,28,41,43,44,54,62,64,71,72,102, 106,107,112,113]	[2,10,19,20,23,24,27,30,34,37,39,42,50, 57,65,73,76,77,83,90,93–101,108,110, 111,114,115]	
Aggregation of participant energy loads	[20,28,35,69,111,116]	[35]	
Social benefits	[28,102,113]		
Shared benefits across the community	-	-	[82]
Identification of origin of energy supply	[28,44,71,72,74,107]	-	-
Environmental benefits	[28,102]	-	-
Community as focal point for engagement		-	[18,82]
Participant types and scales			
Consumers with DV solar systems	[112]	[115]	[47 40]
Small scale participants	[112] [20 22 24 21 26 52 52 55 56 58 72 82 01	[115] [24 26 51 65 82 104 115 118]	[47,48]
Sinai-scale participants	105 106 111 112 1171	[24,20,31,03,03,104,113,110]	-
Prosumers	[1,22,24,25,28,31,44,54,56,62,64,68,85,87, 91,117,119]	[22,24,30,42,51,73,78,83,87,104]	-
Equal in size market participants	[20,41,54,106]	[65]	-
Diverse types of energy generators and consumers	[20,25,35,38,49,52,55,58,105,111]	[24,28,35,50,51,76,83,110,120]	-
Individuals or communities	[75]	-	-
Groups of participants	[38,102]	-	-
Commercial power consumers	[57]	-	-
Participant interactions			
Participants are independent and self-interested consumers	[41,44,49,61,62]	[24,30]	-
Autonomous communication between participants	[29,85]	[39]	-
Active end user participation	[20,24,43,54,58,60,84,102,113]	[20,22–24,30,51,93]	-
Participants trade freely with one another	[63,106]	-	-
Participants have shared energy trading goals	[102]	-	-
Participants can be added or removed without affecting network	[64]	-	-
Market or system design			
Two-sided market	[20]	[20]	_
Smart IT platform	[30.31.33.38.42-44.60-62.64.68.71.72.103.	2.24.30.39.42.50.51.77.93.102.104.110.	_
<u>.</u>	106,121]	111]	
Market rules and mechanisms	[38,44,52,64]	[2,30,42]	-
Market based price signals	[20,35,43,68]	[10,19,20,35,65,108]	-
Interaction with third parties and wholesale market	[24,54,74,75]	[20,24,78]	-
Dynamic pricing	[41,43,71]	[19,20,70,114]	-
Distributed market coordination and management	[122,123]	[83,104,114,118]	-
Decentralised market coordination and management	[1,20,24,40,61,64,67,69,103,112]	[20,24,34,39,64,65,94]	-
Competitive market structure and pricing	[38,44,62,106,122]	[24,51,83]	-
Centralised market coordination and management	[28,35,68,102]	[30]	-
Autonomous grid control	[1]	[27,39,42]	-
Time of use tariffs	[71]	-	-
Sub market within traditional market	[20,73,75]	-	-
Nodal or local grid pricing	[72]	-	-
Grid as supplier of last resort	[38,52,62,74,75]	-	-
Coordination and management of smart devices	-	[24,39,76]	-
Concept based on sharing economy	[25,75]	-	-
		(continued of	on next page)

15

Table A.9 (continued).

Market system and searc			
Local energy trading or market (geographic region)	[24,28,30,31,33,40,42,52,60,62,67,84,84,87, 103,105,112,113,119,121,124,125]	[24,26,28,30,51,77]	[47,80]
Bound by LV-distribution network	[31]	[115]	[47,126,127]
Virtual energy trading (no geographical restrictions)	[56]	[24 28 47 77 79 83]	_
Operate in grid-connected mode	[24 30 52 64 86]	[34 104]	_
Bound by local community	[40]	[30,34]	_
Bound by distribution network	[22]	[24 65 70 101]	_
Operates on various levels of the grid			_
operates on various revers of the grid		114]	
Operate on public electricity grid	-	-	[47,80]
Operate on private electricity grid	-	-	[47,80]
Operate in island-mode	[64]	-	-
Connected at distribution grid level	-	[70,115]	-
Bound by building or block	-	-	[48]
Market transactions and settlement			
Real-time trading	[40,41,56,72,81,91]	[24,26,34,39,70]	[81]
DLT for energy transaction and management	[35,40,44,68,81,82,106,124]	[26,118]	[81]
Automatic executions and settlements	[18,29,35,69,72,81,82,106]	[2,24,26,39]	[81]
Use of virtual or real currencies	[44,81]	-	[81]
Trading of renewable energy	[25,30,41-44,53,62,69,103,125]	[42]	_
Physically delivered energy	[54]	[70]	_
Multi-directional exchange of energy and value	[20.24.41.67.68.72.81.88.119.121.125]	[27.90.128]	_
Manual energy trading	[35.81]	-	[81]
Frequent transactions	[42.86]	[42,90]	-
Energy transactions without intermediary	[3,18,22,24,25,28,29,35,38,40–42,49,55–57,	[42,64,83,110,115]	-
Energy transactions through intermediant		[20]	
Contracting for anomal contract of delivered	[10,29,03,02,00,07]		-
Contracting for energy services onered and delivered	[1,18,29,34,81,82]		-
Bi-directional energy flows	[62,85]	[21,24,39,73,89]	-
Transactions of reactive power	-	[34]	-
Transactions of real power	-	[34]	-
Trading of different classes of energy	[38,44,49,58,88]	-	-
Multilateral bidding transactions	[40,49,88]	-	-
Forward energy transactions	-	[70]	-
Deferred energy transactions	-	[26]	-
Governance and policy			
Shared ownership of generation assets	[49,64,91,112]	-	[47,48,80,126]
Non-traditional business model	[61,74]	[92,110,115]	-
Assets owned by consumers and third parties	[82]	[21]	-
Participants not selling energy professionally	[54]	-	-
Owners have control over devices	-	[39]	-
Extension of existing business models	[54,75]	-	-
Equal responsibility and rights for all parties	[105,117]	-	-
Community operates as legal entity	-	-	[127]

important for CSC that the community operates and is recognised as a legal entity.

Based on these characteristics, CSC is defined as a communityoriented framework which operates as a legal entity and focuses on creating shared benefits for local communities. The system is typically bound by the local LV network or a small geographical region. Participants are typically small-scale consumers and prosumers, and ownership of generation assets can be shared within the community.

To facilitate the process of identifying which model a project is most similar to, in Appendix B a tool is provided that allows for the classification of projects into P2P, TE or CSC based on characteristics.

6. Conclusion

This paper aimed to answer the following research question :what are the key defining characteristics of energy systems labelled as Peer-to-Peer, Community/Collective Self-Consumption or Transactive Energy, as outlined in the literature and by stakeholders? Findings are based on the results of five expert group interviews, representing a wide range of disciplinary and sectoral perspectives across 13 countries, as well as a systematic and targeted review of 78 academic papers and 55 nonacademic papers. While the expert interviews focused on the shared characteristics between these models, the literature review specifically analysed the characteristics that set these models apart from one another. The common themes identified in the expert interview manifest differently in each individual model as shown in the distinct definitions.

The terms 'peer-to-peer energy trading' and 'transactive energy' are often used interchangeably in the field, since they have many overlapping characteristics. TE tends to be defined more broadly, while P2P provides more detailed characteristics as regards market design and configuration. CSC is defined to a limited extent in the literature. The analysis of the characteristics used to define the three concepts that have resulted in three descriptive definitions of P2P, TE and CSC (see Section 5.2).

Findings from both analyses, the interviews and the literature review, indicating the characteristics that distinguish P2P/TE/CSC from traditional energy markets are as follows: They are sub-markets that operate within or alongside traditional energy markets. They involve a form of energy trading or sharing; rely on some form of automation of transactions; are characterised by their promotion and support of the local generation and consumption of energy; encompass both geographically-bounded trading and non-geographically bounded trading; and involve trading with or without intermediaries, with price negotiation mechanisms that reflect the aims of the market. The key differences between P2P, TE, and CSC stem from the problems they

were designed to tackle and their origins. As these models scale and inevitably interact with new market layers and challenges, they may converge in the future, with distinctions between them becoming a function of local contexts and associated social, technical, and economic constraints.

It should be noted that this analysis was subject to several limitations. First, given the vast amount of literature concerning P2P/TE/ CSC, it was only possible to analyse a snapshot of the literature. As noted in Section 2.2, the academic literature was largely dominated by papers modelling different market parameters. To counteract this over-representation of a particular discipline, a more dichotomous and qualitative approach was taken, looking at whether or not a characteristic was present for each term, as opposed to looking at how frequently characteristics were mentioned for each term. This means that in practice, equal weight was given to characteristics that were mentioned a large number of times and those that were mentioned by just a few papers. This can be seen as a defensible approach; given the discrepancy in the number of papers represented by different disciplines, discussing the frequencies of characteristics would have meant amplifying this over-representation. Finally, characteristics were coded based on the explicit mention of that characteristic in the text. It may be that some characteristics were assumed but not explicitly mentioned by certain models.

Nonetheless, this analysis has clear implications for future research and for policy. Recommendations for future research are the following:

- Expand the scope of research beyond modelling. Detailed information on verification processes and licensing frameworks is yet to be provided in the literature.
- Research should be conducted on the scaling up of such models, including on the socio-economic aspects of these models, for instance governance of trading within CSC.
- Expand research on the practicalities of trading energy within CSC models. Research conducted as part of this paper showed that the literature on this topic is significantly lacking.

Recommendations for policymakers and industry stakeholders are the following:

- Promote and provide a dialogue, particularly between different sectors, on aligning their understanding of P2P/TE/CSC models.
- Ensure that the definitions included in legislation and policies reflect the multi-faceted nature of how these models are defined in practice.
- Realise that these models (and thereby their definitions) are evolving quickly and that the research is expanding. Dialogue between industry, academics and policymakers working in the field is therefore crucial.
- Provide definitions in standards, in order to promote interoperability of systems and ensure the safety of consumers playing an active role in these models.

Since this programme of work began, the European Commission has put forward a proposal to enshrine the right to energy sharing for EU citizens [129]. The Commission has defined energy sharing as "self-consumption by active customers of renewable energy generated or stored offsite either from facilities they own, lease, rent in whole or in part or which has been transferred to them by another active customer". However, it is at the discretion of EU countries to transpose this definition into national regulations, and national regulators may wish to propose more nuanced definitions with regards to specific 'energy sharing' models, such as P2P, TE or CSC. A commonly agreed and more detailed definition of these models and their respective goals could support regulators in developing refined definitions in national regulations.

These findings provide a starting point for developing a shared understanding of this class of models, across multiple disciplinary perspectives and applications. This contributes towards enabling knowledge exchange between sectors and disciplines, supporting international collaboration, and deployment of P2P, TE and CSC at scale.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

The completed differentiation tool including the instructions as shown in Table B.10 is available in the supplementary material, which can be found at https://osf.io/mn7zk/, an open source cloud-based project management platform hosted at OSF (Watson et al., 2020). The supplementary material also contains the full interview guide, the Excel sheet used to calculate the percentage similarity between terms, a full matrix of terms and characteristics associated with one another, and the review search protocol.

Acknowledgements

This publication is part of the work of the Global Observatory on Peer-to-Peer, Community Self-Consumption and Transactive Energy Models (GO-P2P), a Task of the User-Centred Energy Systems Technology Collaboration Programme (UsersTCP), run under the auspices of the International Energy Agency (IEA). GO-P2P benefits from the support of Australia, Belgium, Ireland, Italy, the Netherlands, New Zealand, Switzerland, the United Kingdom and the United States. The authors would like to thank the GO-P2P participants who participated in the expert interviews organised in the summer of 2020.

This work was supported by the EPSRC Centre for Doctoral Training in Energy Demand (LoLo), grant numbers EP/L01517X/1 and EP/H009612/1; as well as the EPSRC grant EnergyREV (Energy Revolution Research Consortium) under grant number EP/S031863/1; and the Centre for Research into Energy Demand Solutions, grant reference number EP/R035288/1.

Appendix A. Overview of characteristics and references

See Table A.9.

Appendix B. Differentiation tool for P2P/TE/CSC models

The analysis in Section 4 has shown that while P2P/TE/CSC models all have some features in common, they also have distinct features. This is important because a coherent understanding of these types of models can avoid misunderstandings in communication between regulators and researchers and thus speed up the adoption of LEMs. More specifically, the identification of different concepts can serve policymakers and regulators as a guide to reflect those differences in regulation and therefore provide more certainty and guidance for pilot project stakeholders. In the following, a screening tool to classify a pilot project as either a P2P, TE or CSC model is presented by calculating a similarity index that indicates whether a project is most similar to one concept over the other. In accordance with the approach taken in this research, the similarity index is calculated by using the characteristics associated with P2P, TE, and CSC. The tool is available in the supplementary material of this paper. The differentiation tool was developed by creating a list of all 87 characteristics identified in Section 4.2 that were presented in the categories of 'market or system aims and incentives', 'participant types and scale', 'participant interactions', 'market or system design', 'market or system scale', 'market transactions and settlement' and 'governance and policy'. In order to reduce the size and improve the comprehension of the list of characteristics, the list was screened a second time and the characteristics were assessed for their meaningfulness and uniqueness.

Table B.10

Overview of differentiation tool with example data to calculate similarity index.

Characteristic	Present in pilot	P2P	TE	CSC
Priority on system optimisation		х	х	
Priority on individual benefits	х	х		
Priority on community benefits	х			х
Involves participants of different scales		х		
Focus on small-scale non-commercial individual users	х	х	х	х
Focus on communities (optimisation and metering on community level)				х
Participants are independent and self-interested decision makers	х	х	х	
Active end-user participation	х	х		
Participants have shared energy trading goals				х
Distributed market coordination and management		х	х	
Centralised market coordination and management	х	х	x	
Autonomous grid control			x	
Optimisation at device level			x	
Concept based on sharing economy		х		х
Market-based approach		х	х	
Incentivise user behaviours through price signals		х	х	
Local energy trading or market geographic region	х	х	х	
Virtual energy trading no geographical restrictions		х	x	
Operates on various levels of the grid			x	
Bound by building or block				х
Multi-directional exchange of energy and value	х	х	х	
Energy transactions without intermediary		х	x	
Energy transactions through intermediary		х	x	х
Trading of different classes/attributions of energy	х	х		
Trading of different types of energy (reactive, active)			х	
Participants not selling energy professionally	х	х		
Equal responsibility and rights for all participants		х		
Community operates as legal entity				х
Disruptive business model		х		x
No of characteristics	10	21	17	10
No of shared characteristics		9	4	3
Similarity index		0.43	0.24	0.3

Both coders reviewed the list independently, and if a rewording or further description of a characteristic was deemed necessary, the coders discussed any differences and agreed on an updated version of the list. This resulted in the list being reduced to a total of 30 characteristics. 57 characteristics were either removed or merged with others for clarity. The differentiation tool is presented in a table-based format, as shown in Table B.10. In the first column, the key characteristics are presented. In the second column the user can indicate the characteristics that they think are represented in the particular pilot project. The tool is highly dependent on the expertise of the user, and therefore only serves as an indication for the classification of a pilot project. The characteristics entered can then be compared with P2P, TE, and CSC models using columns three to five. The similarity index is calculated by comparing how many of the characteristics found in a pilot project overlap with the characteristics associated with one of the concepts. In the case of the example data used in Table B.10, the pilot project would be most similar to the concept of P2P and least similar to TE, with CSC in between both concepts. It is at the discretion of the user to interpret the results considering all additional contextual and environmental information available on the pilot project. It is recommended to consider both the characteristics presented and the index value calculated to develop an understanding of the type of concept represented in a pilot project. Similarly to the above-presented analyses, this tool is also impacted by the underrepresentation of CSC in the field. In comparison to the concepts P2P and TE, CSC has fewer characteristics associated with it, therefore weakening the informative value of the tool for this concept in particular. The same limitations as the literature review analysis also apply to this tool (see Section 6). As more institutional bodies or standards organisations continue their work on defining and differentiating P2P/TE/CSC models, better and more accurate characteristics can be derived. These refined characteristics can be used to replace the current list of characteristics and thus contribute to an improved version of the differentiation tool, which will lead to more accurate and more widely applicable results.

References

- Parag Y, Sovacool BK. Electricity market design for the prosumer era. Nat Energy 2016;1:1–6. http://dx.doi.org/10.1038/nenergy.2016.32, URL: https: //vandebron.nl.
- [2] Thomas S, Surampudy M, Knight M. Transactive energy: real-world applications for the modern grid. Technical report, Smart electric power alliance; 2019, URL: https://sepapower.org/resource/transactive-energy-realworld-applications-for-the-grid/.
- [3] IRENA. Peer-to-peer electricity trading: innovation landscape brief. Technical report, Abu Dhabi: International Renewable Energy Agency (IRENA); 2020, URL: https://www.irena.org/-/media/Files/IRENA/Agency/Publication/2020/ Jul/IRENA_Peer-to-peer_trading_2020.pdf.
- [4] Mengelkamp E, Notheisen B, Beer C, Dauer D, Weinhardt C. A blockchain-based smart grid: towards sustainable local energy markets. Comput Sci - Res Dev 2017;33:207–14. http://dx.doi.org/10.1007/s00450-017-0360-9.
- [5] Energy regulation sandbox: Guidance for innovators. Ofgem; 2020.
- [6] What is a regulatory sandbox?. Technical report, Ofgem; 2018, URL: https://www.ofgem.gov.uk/publications/what-regulatory-sandbox#:: text=Where%20our%20regulation%20prevents%20the,of%20the%20usual% 20rules%20applying.
- [7] De Almeida L, Cappelli V, Klausmann N, Van Soest H. Peer-to-peer trading and energy community in the electricity market-analysing the literature on law and regulation and looking ahead to future challenges. EUI working papers RSC 2021/35, 2021.
- [8] Gorbatcheva A, Watson N, Khadem S, Schneiders A, Shipworth D. Global perspectives on peer-to-peer, community self-consumption & transactive energy models. Technical report, ICEEE; 2022.
- [9] European Commission. Directive (EU) 2018/2001 (recast) on the promotion of the use of energy from renewable sources. Off J Eur Union 2018.
- [10] GridWise transactive energy framework version 1.0. GridWise Architecture Council; 2015.
- [11] O'regan B, Silva F, O'leidhin E, Tahir F, Mould K, Hayes B, Hosseinnezhad V, Chitchyan R, Lyons P. P2p, csc and te: A survey on hardware, software and data. Energies 2021;14. http://dx.doi.org/10.3390/en14133851.
- [12] Dudjak V, Neves D, Alskaif T, Khadem S, Pena-Bello A, Saggese P, Bowler B, Andoni M, Bertolini M, Zhou Y, Lormeteau B, Mustafa MA, Wang Y, Francis C, Zobiri F, Parra D, Papaemmanouil A. Impact of local energy markets integration in power systems layer: A comprehensive review. Appl Energy 2021;301. http://dx.doi.org/10.1016/j.apenergy.2021.117434.

- [13] Capper T, Gorbatcheva A, Mustafa MA, Bahloul M, Schwidtal JM, Chitchyan R, Andoni M, Robu V, Montakhabi M, Scott IJ, Francis C, Mbavarira T, Espana JM, Kiesling L. Peer-to-peer, community self-consumption, and transactive energy: A systematic literature review of local energy market models. 2022, http: //dx.doi.org/10.1016/j.rser.2022.112403.
- [14] Adams S, Brown D, Cárdenas Álvarez JP, Chitchyan R, Fell MJ, Hahnel UJ, Hojckova K, Johnson C, Klein L, Montakhabi M, Say K, Singh A, Watson N. Social and economic value in emerging decentralized energy business models: A critical review. 2021, http://dx.doi.org/10.3390/en14237864.
- J. Humbley C Laurén, editor. Languages for special purposes. De Gruyter; 2018, http://dx.doi.org/10.1515/9783110228014.
- [16] ISO/TR 22134:2007 practical guidelines for socioterminology. ISO/TR 22134:2007, International Organization for Standardization; 2007.
- [17] Halperin S, Heath O. Political research: methods and practical skills. 3rd ed., 2020.
- [18] Frieden D, Tuerk A, Roberts J, D'Herbemont S, Gubina A. Collective self-consumption and energy communities: overview of emerging regulatory approaches in europe. Technical report, COMPILE: Integrating community power in energy islands; 2019, URL: https://www.compile-project.eu/wpcontent/uploads/COMPILE_Collective_self-consumption_EU_review_june_2019_ FINAL-1.pdf.
- [19] Council TGA. Decision-maker's transactive energy checklist. 2016.
- [20] Khorasany M, Razzaghi R, Dorri A, Jurdak R, Siano P. Paving the path for two-sided energy markets: An overview of different approaches. IEEE Access 2020;8:223708–22. http://dx.doi.org/10.1109/ACCESS.2020.3040391.
- [21] Bhattarai BP, Hammerstrom DJ, Wang P. Definition of market performance metrics to enable design and valuation of retail transactive energy system markets. In: 2018 IEEE conference on technologies for sustainability. SusTech 2018, Institute of Electrical and Electronics Engineers Inc.; 2019, http://dx.doi. org/10.1109/SusTech.2018.8671375.
- [22] Khorasany M, Mishra Y, Ledwich G. Hybrid trading scheme for peer-topeer energy trading in transactive energy markets. IET Gener Transm Distrib 2020;14:245–53. http://dx.doi.org/10.1049/iet-gtd.2019.1233.
- [23] Adebisi B. Transactive energy systems. In: Proceedings of the 2nd international conference on future networks and distributed systems. New York, NY, USA: Association for Computing Machinery (ACM); 2018, p. 1. http://dx.doi.org/ 10.1145/3231053.3231131, URL: https://dl.acm.org/doi/10.1145/3231053. 3231131.
- [24] Abrishambaf O, Lezama F, Faria P, Vale Z. Towards transactive energy systems: An analysis on current trends. Energy Strategy Rev 2019;26:100418. http: //dx.doi.org/10.1016/j.esr.2019.100418.
- [25] Zhou Y, Wu J, Long C, Ming W. State-of-the-art analysis and perspectives for peer-to-peer energy trading. 2020, http://dx.doi.org/10.1016/j.eng.2020.06. 002.
- [26] Zia MF, Benbouzid M, Elbouchikhi E, Muyeen SM, Techato K, Guerrero JM. Microgrid transactive energy: Review, architectures, distributed ledger technologies, and market analysis. 2020, http://dx.doi.org/10.1109/ACCESS.2020. 2968402.
- [27] Electric power technical whitepaper building a robust value mechanism to facilitate transactive energy. LO3 Energy; 2017.
- [28] Gall N. Trading sunlight: prospects for peer to peer energy trading in the UK solar industry. Technical report, London, UK: Solar Trade Association; 2019, URL: https://www.solar-trade.org.uk/wp-content/uploads/2019/ 12/STA-report-WEB.pdf.
- [29] Maš M, Nina K, Jacob B, Nancy B. New Clean Energy Communities in a Changing European Energy System (NEWCOMERS): Deliverable D2.2 developed as part of the NEWCOMERS project, funded under EU H2020 grant agreement 837752. 2019, URL: https://www.newcomersh2020.eu/upload/files/D2_ 2_newcomers_typology_of_new_clean_energy_communities.pdf.
- [30] Tushar W, Yuen C, Saha TK, Morstyn T, Chapman AC, Alam MJE, Hanif S, Poor HV. Peer-to-peer energy systems for connected communities: A review of recent advances and emerging challenges. Appl Energy 2021;282:116131. http://dx.doi.org/10.1016/j.apenergy.2020.116131.
- [31] Scuri S, Tasheva G, Barros L, Nunes NJ. An HCI perspective on distributed ledger technologies for peer-to-peer energy trading. In: HCI computer science. LNCS, vol. 11748, Springer Verlag; 2019, p. 91–111. http://dx.doi.org/10. 1007/978-3-030-29387-1_6.
- [32] Denktas B, Pekdemir S, Soykan G. Peer to peer business model approach for renewable energy cooperatives. In: 7th international IEEE conference on renewable energy research and applications. ICRERA 2018, Institute of Electrical and Electronics Engineers Inc.; 2018, p. 1336–9. http://dx.doi.org/10.1109/ICRERA. 2018.8566735.
- [33] Gui EM, MacGill I. Typology of future clean energy communities: An exploratory structure, opportunities, and challenges. Energy Res Soc Sci 2018;35:94–107. http://dx.doi.org/10.1016/j.erss.2017.10.019.
- [34] Hertz-Shargel B, Livingston D. Assessing blockchain's future in transactive energy. Technical report, Washington, DC: Atlantic Council; 2019, URL: https://atlanticcouncil.org/wp-content/uploads/2019/09/BLOCKCHAIN-0919-WEB.pdf.

- [35] Dütsch G, Steinecke N. Use cases for blockchain technology in energy & commodity trading. Technical report, PricewaterhouseCoopers; 2017, URL: http://www.pwc.fr/fr/assets/files/pdf/2017/09/pwc-thought-leadership_ use-casesfor-blockchain-technology-in-energy-and-commodity-trading.pdf.
- [36] Schreck S, Schreck S, Prieur De La Comble I, Prieur De La Comble I, Thiem S, Niessen S, Niessen S. A methodological framework to support load forecast error assessment in local energy markets. IEEE Trans Smart Grid 2020;11:3212–20. http://dx.doi.org/10.1109/TSG.2020.2971339.
- [37] Good N, Martínez Ceseña EA, Mancarella P. Ten questions concerning smart districts. Build Environ 2017;118:362–76. http://dx.doi.org/10.1016/j.buildenv. 2017.03.037.
- [38] Morstyn T, Farrell N, Darby SJ, McCulloch MD. Using peer-to-peer energytrading platforms to incentivize prosumers to form federated power plants. 2018, http://dx.doi.org/10.1038/s41560-017-0075-y.
- [39] Chen S, Liu CC. From demand response to transactive energy: state of the art. J Mod Power Syst Clean Energy 2017;5:10-9. http://dx.doi.org/10.1007/ s40565-016-0256-x, URL: https://link.springer.com/article/10.1007/s40565-016-0256-x.
- [40] Meng S, Sun W, Han D, Zhang C. Decentralized electricity transaction mechanism in distribution network based on blockchain technology. In: Asia-Pacific power and energy engineering conference. APPEEC, vol. 2019-Decem, IEEE Computer Society; 2019, http://dx.doi.org/10.1109/APPEEC45492.2019. 8994519.
- [41] Tushar W, Yuen C, Mohsenian-Rad H, Saha T, Poor HV, Wood KL. Transforming energy networks via peer-to-peer energy trading: The potential of gametheoretic approaches. IEEE Signal Process Mag 2018;35:90–111. http://dx.doi. org/10.1109/MSP.2018.2818327.
- [42] Jogunola O, Ikpehai A, Anoh K, Adebisi B, Hammoudeh M, Son S-Y, Harris G. State-of-the-art and prospects for peer-to-peer transaction-based energy system. Energies 2017;10:2106. http://dx.doi.org/10.3390/en10122106, URL: http: //www.mdpi.com/1996-1073/10/12/2106.
- [43] Hackbarth A, Löbbe S. Attitudes, preferences, and intentions of german households concerning participation in peer-to-peer electricity trading. Energy Policy 2020;138:111238. http://dx.doi.org/10.1016/j.enpol.2020.111238.
- [44] Ahl A, Yarime M, Tanaka K, Sagawa D. Review of blockchain-based distributed energy: Implications for institutional development. 2019, http://dx.doi.org/10. 1016/j.rser.2019.03.002.
- [45] Cornélusse B, Savelli I, Paoletti S, Giannitrapani A, Vicino A. A community microgrid architecture with an internal local market. Appl Energy 2019;242:547–60. http://dx.doi.org/10.1016/j.apenergy.2019.03.109.
- [46] Wolsink M. Framing in renewable energy policies: A glossary. Energies 2020;13:2871. http://dx.doi.org/10.3390/en13112871, URL: https://www. mdpi.com/1996-1073/13/11/2871.
- [47] Lettner G, Auer H, Fleischhacker A, Schwabeneder D, Dallinger B, Moisl F, Roman E, Velte D, Huidobro A. Existing and future PV prosumer concepts. Technical report, Vienna, Austria: PVP4Grid; 2018, URL: https://www.pvp4grid.eu/wp-content/uploads/2018/12/D2.1_Existingfuture-prosumer-concepts_PVP4G-1.pdf.
- [48] Smarter Together. Report on collective self-consumption of photovoltaic. Technical report, Smarter Together; 2016, URL: https://www.smarter-together.eu/ file-download/download/public/429.
- [49] Sousa T, Soares T, Pinson P, Moret F, Baroche T, Sorin E. Peer-to-peer and community-based markets: A comprehensive review. 2019, http://dx.doi.org/ 10.1016/j.rser.2019.01.036.
- [50] Akter MN, Mahmud MA, Haque ME, Oo AMT. Design of coordination mechanisms for price anticipatory transactive microgrids. IET Gener Transm Distrib 2020;14:1298–310. http://dx.doi.org/10.1049/iet-gtd.2019.0953, URL: https: //onlinelibrary.wiley.com/doi/10.1049/iet-gtd.2019.0953.
- [51] Khorasany M, Mishra Y, Ledwich G. Market framework for local energy trading: a review of potential designs and market clearing approaches. IET Gener Transm Distrib 2018;12:5899–908. http://dx.doi.org/10.1049/iet-gtd.2018.5309, URL: https://onlinelibrary.wiley.com/doi/10.1049/iet-gtd.2018.5309.
- [52] Mengelkamp E, Diesing J, Weinhardt C. Tracing local energy markets: A literature review. IT - Inf Technol 2019;61:101–10. http://dx.doi.org/10.1515/ itit-2019-0016, URL: https://www.degruyter.com/document/doi/10.1515/itit-2019-0016/html.
- [53] Montakhabi M, Zobiri F, Van Der Graaf S, Deconinck G, Orlando D, Vanhove S, Callaerts R, Ballon P, Mustafa MA. New roles in peer-to-peer electricity markets: Value network analysis. In: 6th IEEE international energy conference. ENERGYCon 2020, Institute of Electrical and Electronics Engineers Inc.; 2020, p. 389–94. http://dx.doi.org/10.1109/ENERGYCon48941.2020.9236480.
- [54] van Soest H. Peer-to-peer electricity trading: A review of the legal context. Compet Regul Netw Ind 2018;19:180–99. http://dx.doi.org/10. 1177/1783591719834902, URL: http://journals.sagepub.com/doi/10.1177/ 1783591719834902.
- [55] Di Silvestre ML, Gallo P, Ippolito MG, Riva Sanseverino E, Sciume G, Zizzo G. An energy blockchain, a use case on tendermint. In: Proceedings - 2018 IEEE international conference on environment and electrical engineering and 2018 IEEE industrial and commercial power systems europe, EEEIC/i and CPS europe 2018. Institute of Electrical and Electronics Engineers Inc.; 2018, http://dx.doi.org/10.1109/EEEIC.2018.8493919.

- [56] Lavrijssen S, Carrillo Parra A. Radical prosumer innovations in the electricity sector and the impact on prosumer regulation. Sustainability 2017;9:1207. http: //dx.doi.org/10.3390/su9071207, URL: http://www.mdpi.com/2071-1050/9/ 7/1207.
- [57] Kilkki O, Repo S, Mendes G, Ferreira JR, Lezama F, Faia R, Faria P, Vale Z, Matos L, Landeck J, Klein L, Almeida B, Ahmad Y, Terras JM. D2.3 scalable local energy market architecture (second release). Technical report, DOMINOES, Empower, CNET, ISEP, VPS, EDPD; 2019, URL: https://ec.europa.eu/research/participants/documents/downloadPublic? documentIds=080166e5cad9d2d1&appId=PPGMS.
- [58] Moret F, Pinson P, Papakonstantinou A. Heterogeneous risk preferences in community-based electricity markets. European J Oper Res 2020;287:36–48. http://dx.doi.org/10.1016/j.ejor.2020.04.034.
- [59] Lee W, Kim D, Jin Y, Park M, Won D. Optimal operation strategy for community-based prosumers through cooperative P2P trading. In: 2019 IEEE milan powerTech, powerTech 2019. Institute of Electrical and Electronics Engineers Inc.; 2019, http://dx.doi.org/10.1109/PTC.2019.8810565.
- [60] Crisp T, Kruja K. Future for all: making a future retail energy market work for everyone. Technical report, Citizens Advice; 2019, URL: https://www.citizensadvice.org.uk/Global/CitizensAdvice/Energy/Futureforall_ FINALVersion.pdf.
- [61] Shrestha A, Bishwokarma R, Chapagain A, Banjara S, Aryal S, Mali B, Thapa R, Bista D, Hayes BP, Papadakis A, Korba P. Peer-to-peer energy trading in micro/mini-grids for local energy communities: A review and case study of nepal. 2019, http://dx.doi.org/10.1109/ACCESS.2019.2940751.
- [62] Mustafa MA, Cleemput S, Abidin A. A local electricity trading market: Security analysis. In: IEEE PES innovative smart grid technologies conference europe. IEEE Computer Society; 2016, http://dx.doi.org/10.1109/ISGTEurope. 2016.7856269.
- [63] Mantegazzini BA. Guidelines for the implementation of new business models final version. Technical report, NemoGrid; 2019, URL: http://nemogrid.eu/wpcontent/uploads/Guidelines-for-business-models_final.pdf.
- [64] Tushar W, Member S, Saha TK, Yuen C, Smith D, Vincent Poor H. Peer-to-peer trading in electricity networks: an overview. Technical report, 2020.
- [65] Khorasany M, Azuatalam D, Glasgow R, Liebman A, Razzaghi R. Transactive energy market for energy management in microgrids: The monash microgrid case study. Energies 2020;13:2010. http://dx.doi.org/10.3390/en13082010, URL: https://www.mdpi.com/1996-1073/13/8/2010.
- [66] Mohamed MA, Jin T, Su W. Multi-agent energy management of smart islands using primal-dual method of multipliers. Energy 2020;208:118306. http://dx. doi.org/10.1016/j.energy.2020.118306.
- [67] Deconinck G, Vankrunkelsven F. Digitalised, decentralised power infrastructures challenge blockchains. Proc Inst Civ Eng - Smart Infrastruct Constr 2020;1–11. http://dx.doi.org/10.1680/jsmic.20.00013.
- [68] Regen. Local supply: options for selling your energy locally. Technical report, Regen, The Innovation Centre; 2015, URL: https://www.stephens-scown.co.uk/ app/uploads/2015/07/LocalEnergySupply.pdf.
- [69] Ableitner L, Tiefenbeck V, Meeuw A, Wörner A, Fleisch E, Wortmann F. User behavior in a real-world peer-to-peer electricity market. Appl Energy 2020;270:115061. http://dx.doi.org/10.1016/j.apenergy.2020.115061.
- [70] ABB. Transactive energy: a market-based approach to distribution optimization. Technical report, Raleigh, NC: ABB; 2013, URL: https: //library.e.abb.com/public/2adb614a11ab082c85257c1d006ef2aa/ABB-782-WPO_Transactive_Energy.pdf.
- [71] Brown D, Hall S, Davis ME. Prosumers in the post subsidy era: an exploration of new prosumer business models in the UK. Energy Policy 2019;135:110984. http://dx.doi.org/10.1016/j.enpol.2019.110984.
- [72] Johnston J. Peer-to-peer energy matching: Transparency, choice, and locational grid pricing. In: Innovation and disruption at the grid's edge: how distributed energy resources are disrupting the utility business model. Elsevier Inc.; 2017, p. 319–30. http://dx.doi.org/10.1016/B978-0-12-811758-3.00016-4.
- [73] Siano P, De Marco G, Rolan A, Loia V. A survey and evaluation of the potentials of distributed ledger technology for peer-to-peer transactive energy exchanges in local energy markets. IEEE Syst J 2019;13:3454–66. http://dx.doi.org/10. 1109/JSYST.2019.2903172.
- [74] Klaassen E, Van der Laan M. Energy and flexibility services for citizens energy communities. Technical report, USEF foundation; 2019, URL: https://www.nweurope.eu/media/6768/usef-white-paper-energy-andflexibility-services-for-citizens-energy-communities-final-cm.pdf.
- [75] CEER. CEER consultation on dynamic regulation to enable digitalisation of the energy system. Technical report, Council of European Energy Regulators (CEER); 2019, URL: https://www.ceer.eu/documents/104400/-/-/3aedcf03-361b-d74f-e433-76e04db24547.
- [76] Cazalet E. EMIX note: transactive energy. Technical report, EMIX Transactive Energy Committee Note; 2010, URL: https://www.oasis-open.org/committees/ download.php/43584/DraftEMIXNoteonTransactiveEnergy.pdf.
- [77] World Energy Council. World energy insights brief: is blockchain in energy driving an evolution or a revolution?. Technical report, London, UK: World Energy Council and PricewaterhouseCoopers; 2018, URL: https://www.pwc.com/gx/ en/energy-utilities-mining/pdf/blockchain-anthology-of-interviews-2018.pdf.

- [78] Georgilakis PS. Review of computational intelligence methods for local energy markets at the power distribution level to facilitate the integration of distributed energy resources: State-of-the-art and future research. Energies 2020;13:186. http://dx.doi.org/10.3390/en13010186, URL: https://www.mdpi.com/1996-1073/13/1/186.
- [79] STEP. Transactional energy management for residential buildings. Technical report, Toronto, Canada: Toronto and Region Conservation Authority's Sustainable Technologies Evaluation Program (STEP); 2018, URL: https://sustainabletechnologies.ca/app/uploads/2018/07/Transactional-Energy-Buildings_Tech-Brief.pdf.
- [80] Fleischhacker A, Radl J, Lettner G, Schwabeneder D, Moisl F, Auer H. Improvement of PVP4grid concepts. 2018.
- [81] Donnerer D, Lacassagne S. Blockchains and energy transitions what challenges for cities?. Technical report, Energy Cities; 2019, URL: https://energy-cities. eu/wp-content/uploads/2019/01/energy-cities-blockchain-study_2018_en.pdf.
- [82] Schittekatte T, Reif V, Nouicer A, Meeus L. TSO-dSO-consumer INTERFACE architecture to provide innovative grid services for an efficient power system. Technical Report, INTERRFACE Consortium; 2020, URL: https://youngfoundation.org/wp-content/uploads/2017/02/Share-to-Knowsummary-guide.pdf.
- [83] Abdella J, Shuaib K. Peer to peer distributed energy trading in smart grids: A survey. Energies 2018;11:1560. http://dx.doi.org/10.3390/en11061560, URL: http://www.mdpi.com/1996-1073/11/6/1560.
- [84] Ableitner L, Meeuw A, Schopfer S, Tiefenbeck V, Wortmann F, Wörner A. Quartierstrom – Implementation of a real world prosumer centric local energy market in Walenstadt, Switzerland. 2019, arXiv URL: http://arxiv.org/abs/ 1905.07242.
- [85] Jogunola O, Ikpehai A, Anoh K, Adebisi B, Hammoudeh M, Gacanin H, Harris G. Comparative analysis of P2P architectures for energy trading and sharing. Energies 2017;11:62. http://dx.doi.org/10.3390/en11010062, URL: http:// www.mdpi.com/1996-1073/11/1/62.
- [86] Peer-to-Peer distributed ledger technology assessment Virtual peer-to-peer energy trading using distributed ledger technology: comprehensive project assessment report. AGL; 2017.
- [87] Zhou Y, Wu J, Long C, Cheng M, Zhang C. Performance evaluation of peer-topeer energy sharing models. In: Energy procedia. vol. 143, Elsevier Ltd; 2017, p. 817–22. http://dx.doi.org/10.1016/j.egypro.2017.12.768.
- [88] Baroche T, Pinson P, Latimier RLG, Ben Ahmed H. Exogenous cost allocation in peer-to-peer electricity markets. IEEE Trans Power Syst 2019;34:2553–64. http://dx.doi.org/10.1109/TPWRS.2019.2896654.
- [89] Hou W, Tian G, Guo L, Wang X, Zhang X, Ning Z. Cooperative mechanism for energy transportation and storage in internet of energy. IEEE Access 2017;5:1363–75. http://dx.doi.org/10.1109/ACCESS.2017.2664981.
- [90] QualityLogic. Transactive control and OpenADR roles and relationships. 2014.
- [91] Azim MI, Tushar W, Saha TK. Investigating the impact of P2P trading on power losses in grid-connected networks with prosumers. Appl Energy 2020;263:114687. http://dx.doi.org/10.1016/j.apenergy.2020.114687.
- [92] LO3 Energy. Introduction to exergy. Technical report, LO3 Energy; 2018, URL: https://lo3energy.com/wp-content/uploads/2018/04/Exergy-BIZWhitepaperv11.pdf.
- [93] Atamturk N, Zafar M. Transactive energy: a surreal vision or a necessary and feasible solution to grid problems?. Technical report, California: California Public Utilities Commission; 2014, URL: http://www.cpuc.ca. gov/uploadedFiles/CPUC_Public_Website/Content/About_Us/Organization/ Divisions/Policy_and_Planning/PPD_Work/PPDTransactiveEnergy_30Oct14.pdf.
- [94] Cazalet E, De Martini P, Price J, Woychik E, Caldwell J. Transactive energy models. 2016.
- [95] Lian J, Marinovici L, Zhang W, Kalsi K, Sun Y, Widergren SE. Transactive system part i: theoretical underpinnings of payoff functions, control decisions, information privacy, and solution concepts. Technical report, Oak Ridge, Tennessee: U.S. Department of Energy under Contract; Pacific Northwest National Laboratory; 2017, URL: https://www.google.com/url?q=https: //www.pnnl.gov/main/publications/external/technical_reports/PNNL-27235Part1.pdf&sa=D&source=editors&ust=1620051724027000&usg= AFQjCNEMCZEhfzj6j3hx71-kHMQjWKFytQ.
- [96] Holmberg DG, Bushby ST. NIST transactive energy modeling and simulation challenge for the smart grid phase I report. 2018.
- [97] Holmberg D, Burns M, Bushby S, Gopstein A, McDermott T, Tang Y, Huang Q, Pratt A, Ruth M, Ding F, Bichpuriya Y, Rajagopal N, Ilic M, Jaddivada R, Neema H. NIST transactive energy modeling and simulation challenge phase II final report. 2019.
- [98] Hammerstrom DJ. History and implications of emerging transactive energy systems. Technical report, Tampa, Florida: U.S. Department of Energy under Contract; Pacific Northwest National Laboratory; 2017, URL: https://www.psma.com/sites/default/files/uploads/tech-forums-alternativeenergy/presentations/is84-history-and-implications-emerging-transactiveenergy-systems.pdf.

- [99] Deloitte. Choose, aggregate, transact: increasing options for electricity customers. Technical report, Deloitte Center for Energy Solutions; 2017, URL: https://www2.deloitte.com/content/dam/Deloitte/global/Documents/Energyand-Resources/gx-energy-and-resources-increasing-options-for-electricitycustomers.pdf.
- [100] Caldwell J, Cazalet E, Cox B, De Martini P, Dobriansky L, Heitman P, Price J, Radloff G, Rahimi F, Woychik E. The policymaker's toolkit: Vital questions to be addressed about proposed transactive energy systems. Technical report, NIST National Institute of Standards and Technology; 2017, URL: https://s3.amazonaws.com/nist-sgcps/TEChallenge/Library/ BusRegTeam-PolicymakerToolkitPaper-2017.pdf.
- [101] Guerrero J, Gebbran D, Mhanna S, Chapman AC, Verbič G. Towards a transactive energy system for integration of distributed energy resources: Home energy management, distributed optimal power flow, and peer-to-peer energy trading. 2020, http://dx.doi.org/10.1016/j.rser.2020.110000.
- [102] Espe E, Potdar V, Chang E. Prosumer communities and relationships in smart grids: A literature review, evolution and future directions. Energies 2018;11:2528. http://dx.doi.org/10.3390/en11102528, URL: http://www. mdpi.com/1996-1073/11/10/2528.
- [103] Montakhabi M, Madhusudan A, Van Der Graaf S, Abidin A, Ballon P, Mustafa MA. Sharing economy in future peer-to-peer electricity trading markets: Security and privacy analysis. In: Workshop on decentralized ioT systems and security. DISS, vol. 2020, 2020, http://dx.doi.org/10.14722/diss.2020.23005.
- [104] Klein LP, Matos LM, Allegretti G. A pragmatic approach towards enduser engagement in the context of peer-to-peer energy sharing. Energy 2020;205:118001. http://dx.doi.org/10.1016/j.energy.2020.118001.
- [105] Khan I. Drivers, enablers, and barriers to prosumerism in Bangladesh: A sustainable solution to energy poverty? Energy Res Soc Sci 2019;55:82–92. http://dx.doi.org/10.1016/j.erss.2019.04.019.
- [106] Blockchain innovation landscape brief. IRENA; 2019.
- [107] Schneiders A, Fell MJ, Nolden C. Peer-to-peer electricity trading and the sharing economy: social, markets and regulatory perspectives. 2022, http:// dx.doi.org/10.1080/15567249.2022.2050849, URL: https://www.tandfonline. com/doi/abs/10.1080/15567249.2022.2050849.
- [108] Patterson BT, Geary DE. Real-time transactional power management in a microgrid mesh network: The enernet. In: INTELEc, international telecommunications energy conference (proceedings). vol. 2016-Novem, Institute of Electrical and Electronics Engineers Inc.; 2016, http://dx.doi.org/10.1109/ INTLEC.2016.7749106.
- [109] Akter M, Mahmud M, Oo A. A hierarchical transactive energy management system for energy sharing in residential microgrids. Energies 2017;10:2098. http: //dx.doi.org/10.3390/en10122098, URL: http://www.mdpi.com/1996-1073/ 10/12/2098.
- [110] Sharon T. Evolution of the distribution system the potential for distributionlevel markets: A primer for state utility regulators. 2018.
- [111] Ofgem. Ofgem's future insights series: flexibility platforms in electricity markets. Technical report, Ofgem; 2019, URL: Ofgem' sFutureInsightsSeriesFlexibilityPlatformsinelectricitymarkets.
- [112] Etukudor C, Couraud B, Robu V, Früh W-G, Flynn D, Okereke C. Automated negotiation for peer-to-peer electricity trading in local energy markets. Energies 2020;13:920. http://dx.doi.org/10.3390/en13040920, URL: https://www. mdpi.com/1996-1073/13/4/920.
- [113] Kumar C. Community energy 2.0 the future role of local energy ownership in the UK. Technical report, London, UK: Green Alliance; 2019, URL: https: //www.green-alliance.org.uk/resources/Community Energy 2.0.pdf.
- [114] openADR Alliance. OpenADR and transactive energy Is there a transactive openadr in our future?, n.d. URL: https://www.openadr.org/assets/docs/3_ openadr_and%20te%20052116%20jm.pdf.
- [115] Klein LP, Krivoglazova A, Matos L, Landeck J, De Azevedo M. A novel peerto-peer energy sharing business model for the Portuguese energy market. Energies 2019;13:125. http://dx.doi.org/10.3390/en13010125, URL: https:// www.mdpi.com/1996-1073/13/1/125.

- [116] Dunlop S, Roesch A. EU-wide solar PV business models: guidlines for implementation. Technical report, SolarPower Europe; 2016, URL: http://resource-platform.eu/files/knowledge/reports/EU-wide-solar-PVbusiness-models-PV-Financing.pdf.
- [117] Liu N, Yu X, Wang C, Li C, Ma L, Lei J. Energy-sharing model with price-based demand Response for Microgrids of Peer-to-Peer Prosumers. IEEE Trans Power Syst 2017;32:3569–83. http://dx.doi.org/10.1109/TPWRS.2017.2649558.
- [118] Mackinnon L, Vrins J, Williams T, Metz A. ENERGY CLOUD 4.0 capturing value through disruptive energy platforms — focus on europe. Technical report, Navigant; 2018, URL: https://www.european-utility-week.com/_media/ Navigant-Energy-Cloud-4-0---Executive-Summary---Europe.pdf.
- [119] González-Romera E, Romero-Cadaval E, Roncero-Clemente C, Ruiz-Cortés M, Barrero-González F, Milanés Montero M-I, Moreno-Muñz A. Secondary control for storage power converters in isolated nanogrids to allow peer-to-peer power sharing. Electronics 2020;9:140. http://dx.doi.org/10.3390/electronics9010140, URL: https://www.mdpi.com/2079-9292/9/1/140.
- [120] Huang Q, McDermott TE, Tang Y, Makhmalbaf A, Hammerstrom DJ, Fisher AR, Marinovici LD, Hardy T. Simulation-based valuation of transactive energy systems. IEEE Trans Power Syst 2019;34:4138–47. http://dx.doi.org/10.1109/ TPWRS.2018.2838111.
- [121] Zhang C, Wu J, Zhou Y, Cheng M, Long C. Peer-to-peer energy trading in a microgrid. Appl Energy 2018;220:1–12. http://dx.doi.org/10.1016/j.apenergy. 2018.03.010.
- [122] Han X, Sun L, Tao Y, Zhao J, Wang G, Yuan D. Distributed energy-sharing strategy for peer-to-peer microgrid system. J Energy Eng 2020;146:04020033. http: //dx.doi.org/10.1061/(ASCE)EY.1943-7897.0000665, URL: http://ascelibrary. org/doi/10.1061/%28ASCE%29EY.1943-7897.0000665.
- [123] Duchesne L, Cotnélusse B, Savelli I. Sensitivity analysis of a local market model for community microgrids. In: 2019 IEEE milan powerTech. PowerTech 2019, Institute of Electrical and Electronics Engineers Inc.; 2019, http://dx.doi.org/ 10.1109/PTC.2019.8810614.
- [124] Hinson S. Electricity grids. Technical report, The House of Commons Library; 2019, URL: http://researchbriefings.files.parliament.uk/documents/CBP-8472/ CBP-8472.pdf.
- [125] Montakhabi M, Van Der Graaf S, Ballon P, Mustafa MA. Sharing beyond peer-to-peer trading: Collaborative (open) business models as a pathway to smart circular economy in electricity markets. In: Proceedings - 16th annual international conference on distributed computing in sensor systems. DCOSS 2020, Institute of Electrical and Electronics Engineers Inc.; 2020, p. 482–9. http://dx.doi.org/10.1109/DCOSS49796.2020.00081.
- [126] Stephant M, Hassam-Ouari K, Abbes D, Labrunie A, Robyns B. A survey on energy management and blockchain for collective self-consumption. In: 2018 7th international conference on systems and control. ICSC 2018, Institute of Electrical and Electronics Engineers Inc.; 2018, p. 237–43. http://dx.doi.org/ 10.1109/ICoSC.2018.8587812.
- [127] Oriol L. Self-consumption framework in France. Technical report, Ministere de la transition ecologique et solidaire; 2018, URL: https://energie-fr-de.eu/ fr/manifestations/lecteur/conference-sur-lautoconsommation-photovoltaiquecadres-reglementaires-et-modeles-daffaires-785.html?file=files/ofaenr/02conferences/2018/180515_conference_pv_autoconsommation/Presentations/02_ Louise_Orio.
- [128] Barrager S, Cazalet E. Transactive energy: a sustainable business and regulatory model for electricity. Baker Street Publishing; 2014.
- [129] Proposal for a regulation of the European parliament and of the council amending regulations (eu) 2019/943 and (eu) 2019/942 as well as directives (eu) 2018/2001 and (eu) 2019/944 to improve the union's electricity market design. European Comission; 2023, URL: https://eur-lex.europa.eu/legalcontent/EN/TXT/?uri=CELEX%3A52023PC0148.